

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Afonso, António; Alves, José; Grabowski, Wojciech; Monteiro, Sofia

# Working Paper Stock and Sovereign Returns Linkages: Time-Varying Causality and Extreme-Quantile Determinants

CESifo Working Paper, No. 11667

**Provided in Cooperation with:** Ifo Institute – Leibniz Institute for Economic Research at the University of Munich

*Suggested Citation:* Afonso, António; Alves, José; Grabowski, Wojciech; Monteiro, Sofia (2025) : Stock and Sovereign Returns Linkages: Time-Varying Causality and Extreme-Quantile Determinants, CESifo Working Paper, No. 11667, CESifo GmbH, Munich

This Version is available at: https://hdl.handle.net/10419/314706

#### Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

#### Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



# WWW.ECONSTOR.EU



# Stock and Sovereign Returns Linkages: Time-Varying Causality and Extreme-Quantile Determinants

António Afonso, José Alves, Wojciech Grabowski, Sofia Monteiro



# Impressum:

CESifo Working Papers ISSN 2364-1428 (electronic version) Publisher and distributor: Munich Society for the Promotion of Economic Research - CESifo GmbH The international platform of Ludwigs-Maximilians University's Center for Economic Studies and the ifo Institute Poschingerstr. 5, 81679 Munich, Germany Telephone +49 (0)89 2180-2740, Telefax +49 (0)89 2180-17845, email office@cesifo.de Editor: Clemens Fuest https://www.cesifo.org/en/wp An electronic version of the paper may be downloaded • from the SSRN website: www.SSRN.com

- from the RePEc website: <u>www.RePEc.org</u>
- from the CESifo website: <u>https://www.cesifo.org/en/wp</u>

# Stock and Sovereign Returns Linkages: Time-Varying Causality and Extreme-Quantile **Determinants**

# Abstract

We employ a cross-quantilogram approach to assess relationships between quantiles of stock returns and sovereign yields, in the U.S. and Germany, in the period 1990-2024. Specifically, we focus on the lowest 5% quantile of stock returns and the highest 5% quantile of bond returns, providing insights into tail dependencies, crucial during market downturns and periods of heightened volatility. We also measure causality in volatilities extending well-known approaches analyzing volatility transmission. We find significant cross-market relationships between U.S. and German stock and bond markets, influenced by economic crises, macroeconomic dynamics, and monetary policy interventions, and financial stress play a crucial role.

JEL-Codes: C320, F210, F370, F420.

Keywords: stock returns, sovereign bond returns, stock-bond relationship, cross-quantilogram, volatility transmission, US, Germany, monetary policy shocks, fiscal stance.

António Afonso University of Lisbon / Portugal aafonso@iseg.ulisboa.pt

José Alves ISEG – School of Economics and Management ISEG – School of Economics and Management University of Lisbon / Portugal jalves@iseg.ulisboa.pt

Wojciech Grabowski Department of Econometric Models and *Forecasts, University of Lodz / Poland* wojciech.grabowski@uni.lodz.pl

Sofia Monteiro ISEG – School of Economics and Management University of Lisbon / Portugal asmonteiro@iseg.ulisboa.pt

# January 2025

This work was supported by the FCT (Fundação para a Ciência e a Tecnologia) [grant number UIDB/05069/2020]. The opinions expressed herein are those of the authors and do not necessarily reflect those of the authors' employers. Any remaining errors are the authors' sole responsibility. Wojciech Grabowski acknowledges the financial support of the National Agency of Academic Exchange, Poland.

## 1. Introduction

The dynamic relation between stock and bond markets has been a cornerstone of financial and economic research, given its profound implications for portfolio diversification, risk management, and economic policy formulation. Understanding the nature of these relationships is particularly critical during periods of financial turbulence, when traditional correlation measures often fail to capture the complex interdependencies between asset classes. This study investigates the dynamic interactions between stock and bond markets in two pivotal economies – the United States (U.S.) and Germany (DE). By examining these relationships both within and across these economies, we aim to shed light on the complex interdependencies that characterize these financial instruments under varying market conditions, monetary policy stances, and fiscal environments.

The motivation for this study steams from the need to comprehend market behavior during extreme financial events, where conventional correlation measures of correlation and causality often fall short. Traditional approaches typically focus on mean relationships or linear dependencies, which may obscure critical insights into tail dependencies – those occurring at the extremes of return distributions. Such tail dependencies are particularly relevant during periods of market turmoil, fiscal stress, or heightened volatility, when the relationships between asset classes can exhibit nonlinear and asymmetric behaviors. To address these limitations, we employ the cross-quantilogram methodology, a novel approach that allows for the analysis of lead-lag relationships between different quantiles of stock returns, bond yield changes, and their volatilities. Specifically, we focus on the lowest 5% quantile of stock returns and the highest 5% quantile of changes in sovereign bond yield, providing insights into tail dependencies that are crucial for understanding market behavior during downturns and periods of financial instability.

In the context of the United States, the stock-bond relationship has been extensively analyzed, with findings frequently indicating a negative correlation during periods of economic uncertainty. This phenomenon, often referred to as the flight-to-quality, where investors reallocate capital from stocks to bonds during periods of economic uncertainty (Connolly et al., 2005) However, the magnitude and nature of this relationship can vary significantly depending on the prevailing economic environment and market sentiment (Baker and Wurgler, 2007). Similarly, in Germany, the stock-bond relationship is shaped not only by both domestic factors but also by broader European economic conditions. As

the economic linchpin of the Eurozone, Germany's financial markets are closely monitored by investors and policymakers alike. Understanding the dynamics of Germany's stock and bond markets not only provides insights into the country's financial stability but also sheds light on the interconnectedness of European markets (Döpke and Pierdzioch, 2006; Johnson and Soenen, 2009).

Beyond within-country dynamics, this study also explores the cross-country relationships between the U.S. and German stock and bond markets. Given the global integration of financial markets, shocks in one market can disseminate to another, with significant implications for global financial stability.. By analyzing the cross-quantilogram relationships, this research aims to uncover how extreme events in one market can transmit to another, thereby enhancing our understanding of global financial contagion and interdependence (Baur and Lucey, 2009). This is particularly relevant in an era of increasing financial globalization, where the spillover effects of monetary policy, fiscal shocks, and financial stress are amplified across borders.

In addition to examining stock and bond market interactions, we incorporate financial stress indicators, monetary policy shocks, macroeconomic conditions and fiscal stance measures to provide a more comprehensive analysis. Financial stress indicators, such as the Composite Indicator of Systemic Stress (CISS), offer valuable insights into the severity of financial crises and their impact on economic activity (Chavleishvili and Kremer, 2023). Monetary policy shocks, identified through high-frequency data around central bank announcements, reveal the effects of unexpected changes in monetary policy on financial markets (Miranda-Agrippino and Ricco, 2021). Furthermore, we consider the real effective exchange rates on bond and asset returns given its significant influence on financial assets dynamics (Bodart and Reding, 1999; Andersen et al., 2007; Fidora et al., 2007; Viceira, 2012; Valchev, 2020). Lastly, fiscal performance, proxied by the cyclically adjusted primary balance (CAPB), plays a critical role in shaping investor perceptions of sovereign risk and market stability (Ardagna, 2009; Afonso and Sousa, 2011; Centinaio et al., 2024). By integrating these factors, we provide a holistic view of the determinants of stock-bond relationships.

It should be added that to the best our knowledge such in-depth analysis of determinants of causality in extreme quantiles of distribution has not been done before. When we try to evaluate the impact of financial stress, monetary policy shocks, fiscal stance and exchange rates on causality in extreme quantiles of distribution of rates of

return, changes in sovereign bond yields, stock market volatilities and bond volatilities, we contribute to the existing literature in stock-bond relationships.

Specifically, our analysis identifies significant patterns in the cross-relationships between stock and bond markets in the U.S. and Germany. In the early 1990s, extreme negative stock returns in the U.S. led to similar movements in Germany, while negative returns in Germany caused opposite movements in the U.S. The 2006-2008 period showed increased interconnectedness during the global financial crisis, and 2018-2020 saw strong bidirectional spillovers coinciding with geopolitical events like the US-China trade war and Brexit.

Further, the European sovereign debt crisis highlighted significant yield spikes and contagion effects, emphasizing the mutual transmission of financial shocks between the U.S. and Germany bond markets. In Germany, the bidirectional relationship between bonds and stocks was positive until the early 21st century, becoming more variable post-2000. During the sovereign debt crisis, heightened volatility caused stocks and bonds to move in opposite directions. In the U.S., the early 2000s showed positive predictability between stock and bond returns, with the 2010s marked by significant cross-asset predictability.

Moreover, exchange rate fluctuations play a moderating role in the short term. When the euro appreciates against the U.S. dollar, causality from the U.S. bond market to the German bond market increases. Conversely, when the euro weakens, causality from the German stock market to the German bond market intensifies. Monetary policy shocks, particularly from the European Central Bank, significantly impact the causality between the German and U.S. bond markets. On the other hand, fiscal primary balances also affects cross-market dynamics, with higher primary balances strengthening the resilience of the German stock market but not the U.S. bond market.

During periods of heightened financial stress, causality from the German stock market to the German bond market decreases, suggesting that German sovereign bonds act as a safe haven. In the U.S., financial stress impacts causality between its bond and stock markets negatively for a one-day lag. Higher financial stress also intensifies causality in volatility.

Our study is organized as follows: Section 2 presents the literature review. Section 3 discusses the employed methodology, data and sources. Section 4 discusses the baseline

results and additional conducted robustness checks. Finally, Section 5 summarizes the main conclusions and policy implications.

## 2. Literature Review

# 2.1. The relationship between bonds and stocks

The linkages between the returns of stocks and bonds have been deeply analyzed in the literature. Specifically, studies have devoted their efforts to understand the existence of co-movements between these two assets, with empirical evidence of correlation changes between stocks and bonds in different timespans in the last century (Shiller and Beltratti, 1992; Gerrits and Yuce, 1999; Ilmanen, 2003; Jammazi et al., 2015).

Macroeconomic fundamentals have exhibited a significant effect on bonds and stocks trajectories (Yang et al., 2009; Duffee, 2023). For instance, the incorporation of risk and productivity growth, and uncertainty measures to analyze parallel patterns between stocks and bonds have been studied in detail for a long time with different conclusions depending on the variables, economic turmoil and financial stress events, as well as the economies under analysis or financial liquidity (Barsky, 1989; Andersson et al., 2007; Beber et al., 2009; Brière et al., 2012; Chiang et al., 2015; Dufour et al., 2017; Lee, 2021). Moreover, stock market prices are found to play a major role in explaining macroeconomic fundamentals, namely, GDP growth, whilst inflation phenomena are little explained by stock markets evolution (Lee, 1992). Yet, the change between positive and negative correlation among stock and bond prices have been justified with increases in price levels, which lowers bond returns, while higher economic activity leads to an increase in stock prices, therefore, justifying the negative correlation between stocks and bonds in light of a macroeconomic analysis on bond-stocks dynamics (Campbell et al., 2020).

On the other hand, monetary policies are also associated with stock markets dynamics. For instance, an expansionary monetary policy through an unanticipated Federal funds rate cut leads to an increase in stock indexes (Bernanke and Kuttner, 2005). Specifically, Bernanke and Kuttner (2005) found, in their analysis, that the predominant influence on stock price movements is attributable to the effects of unexpected monetary policy actions on anticipated excess returns in the U.S. stock market. However, the monetary policy effects on stock markets are found to be heterogeneous depending on stock markets stage, i.e., if the stock market is in a bearish or bullish regime. In detail, by resorting to modified versions of the Markov-switching model, Chen (2007) demonstrates a significant and

negative impact of contractionary monetary policy on stock returns. Furthermore, the findings indicate that the effects of monetary policy are substantially more pronounced during bear-market periods compared to bull-market periods, highlighting two distinct channels through which a tightening monetary policy depresses stock returns: directly lowering returns and increasing the likelihood of transitions to low-return regimes (bear markets).

Furthermore, Grammatikos and Vermeulen (2012) examine the transmission of the 2007-2010 financial and sovereign debt crises to fifteen EMU countries, resorting to daily data from 2003 to 2010 on financial and non-financial stock market indexes. By analyzing stock market returns for three groups within the EMU, namely, North, South, and Small European economies, these authors found that both Northern and Southern European countries experienced significant crisis transmission effects, whereas the smallest countries appeared relatively insulated from international events. Furthermore, the results the authors reach support a robust evidence of crisis transmission to European non-financial sectors from U.S. non-financial sectors, but not for financial sectors, while financial sectors became significantly more sensitive to changes in the spread between Greek and German CDS after the Lehman's collapse, compared to the pre-Lehman Brothers period. However, this increase in sensitivity is much smaller for non-financial sectors. Lastly, prior to the crisis, euro appreciations were associated with declines in European stock markets, a relationship that reversed during the crisis. These conclusions were more recently corroborated by a study conducted by Jammazi et al. (2017).

Beyond the analysis of macroeconomic and monetary policy effects on bond and stock markets, another strand of literature has devoted efforts to comprehend when and why there are, or not, co-movements between the two assets. For instance, Longin and Solnik (2002) investigate the hypothesis that the international equity market correlation increases during volatile periods, a challenging endeavor due to potential spurious relationships between correlation and volatility. They employ extreme value theory to model the tails of multivariate distributions. By deriving the distribution of extreme correlations across various return distributions, the authors provide evidence supporting the rejection of the null hypothesis of multivariate normality for the negative tail, but not for the positive tail. Furthermore, their findings indicate that correlation is more significantly influenced by market trends than by volatility itself, with correlations increasing in bear markets but not in bull markets. Regarding the co-movements between the stock markets of the U.S. and Germany from January 1980 to September 2002, Bonfiglioli and Favero (2005) do not identify a long-term interdependence between the two markets. However, their empirical findings suggest a short-term interdependence and contagion, wherein short-term fluctuations in U.S. stock prices spill over to German stocks. These co-movements are particularly unstable during episodes of high volatility. Expanding their analysis to encompass G7 countries, Kim and In (2007) employ a wavelet correlation analysis. Their empirical findings indicate that the relationship between fluctuations in stock prices and bond yields varies across different countries and is influenced by the time scale considered. Additionally, the wavelet analysis demonstrates that, with the exception of Japan, changes in stock prices and bond yields generally do not exhibit synchronous movements in most G7 nations. Yet, Ferrer et al. (2016) by making also use of wavelet analysis for European economies found similar conclusions to what Kim and In (2017) reached in their study.

However, Ferrer et al. (2019), by investigating the relationship between long-term bond yields and stock market returns using the quantile-on-quantile method (from January 2001 to March 2016), reveal that the connection between interest rates and equity markets is generally positive. Mainly, the strongest correlations are observed during extreme market conditions, particularly when there are significant drops in 10-year Treasury bond yields and a pronounced bearish trend in stock prices.

Nguyen and Javed (2023) introduce generalized autoregressive score mixed frequency data sampling (GAS MIDAS) copula models to examine the dynamic relationship between stock and bond returns, which are essential for portfolio allocation and risk management and to split the analysis into short-term and long-term dependencies, and also to consider asymmetric dependencies at different quantiles. Therefore, while the long-term dependence is influenced by macro-financial factors through a MIDAS regression, the short-term dependence is modeled using a GAS process, highlighting the relevance of this methodological approach under an optimizing portfolio allocation perspective, and enhancing risk management accuracy.

On the other hand, Baur and Lucey (2009) investigate the phenomenon of investors switching between stocks and bonds, proposing definitions and tests for flight-to-quality, flight-from-quality, and cross-asset contagion. By examining these dynamics, the authors explore their results' implications for the financial system. An empirical analysis of eight developed countries, including the U. S., U. K., Germany, and Japan, reveals that such

flights are prevalent during crises and not confined to individual countries but occur simultaneously across multiple nations. This suggests a connection between these flights and cross-country contagion, at the same time, such flights contribute to the resilience of financial markets by offering diversification benefits during periods of heightened uncertainty. Conversely, Adrian et al. (2019) identify a significant and nonlinear relationship between stock and bond returns and historical equity market volatility, as assessed by the Chicago Board Options Exchange Market Volatility Index (VIX). They develop a novel estimator that utilizes variations in cross-sectional returns to capture this relationship. Their analysis reveals that the nonlinear patterns for stocks and bonds are inversely correlated, illustrating a flight-to-safety effect: as volatility escalates from moderate to high levels, expected returns for stocks increase, while those for Treasury bonds decrease. These findings bolster dynamic asset pricing models that propose the risk premium is a nonlinear function of market volatility.

Lastly, intra-week seasonality plays an important role in the relationship between stocks and bonds. For instance, Flannery and Protopapadakis (1988) examined the persistence of intra-week seasonality in stock and bond markets, focusing on three stock indices and Treasury bonds with seven different maturities. Their research reveals that intra-week seasonality remains significant but exhibits varying patterns across both asset classes and among the different bond maturities. Notably, they observe that Monday returns tend to become more negative as bond maturity increases, a trend seen in both stocks and bonds. These results imply that neither institutional factors nor generalequilibrium theories alone can fully account for the observed intra-week seasonal patterns in the securities markets. In a more recent study, Cho et al. (2007) investigate the Monday effect in daily stock index returns using the stochastic dominance criterion, which is a more robust measure compared to the mean comparison methods used in earlier research. As justified by the authors, this criterion enables a clearer economic interpretation of the obtained results. The findings reveal strong evidence of the Monday effect in many instances under this stringent criterion. While the effect has diminished or reversed in the Dow Jones and S&P 500 indices since 1987, it remains pronounced in broader indices like the NASDAQ, Russell 2000, and CRSP (Center for Research in Security Prices) indices.

When assessing the relationship between stock and bond market, as in the case of the analysis for pairs of stock markets and pairs of bond markets, the problem of volatility transmission is also important. Chuliá and Torró (2008) analyzed volatility spillovers among European stock and bond markets, indicating that volatility spillovers in both directions and that the stock-bond trading rules offer profitable returns. Dean et al. (2010) documented asymmetry in the mechanism of volatility transmission between stock and bond market. The results for these two segments of the Australian financial market indicated that in the period 1992-2006 bond market volatility spilled over into the equity market, but the reverse was not true. Moreover, the transmission of bond market volatility into stock market volatility depended on signs of the return shocks in each market.

As Reinhart and Rogoff (2008) indicate, during crises, financial market volatility increases sharply and spills over across markets. Motivated by such considerations Diebold and Yilmaz (2012) introduced a volatility spillover measure based on forecast error decompositions from the VAR model. They analyzed volatility spillover among various asset classes including stocks and bonds. The analysis of volatility spillovers for stock-bond pairs indicated net positive volatility spillovers from stock market to bond market during the Global Financial Crisis, as well as in the years 2002-2003. Hence, this means that positive net transmission from the stock to the bond market was recorded in periods of heightened financial stress, while positive net transmission in the opposite direction was observed in tranquil times. In opposition to Diebold and Yilmaz (2012), Baruník and Křehlík (2018) proposed the analysis of volatility transmission in frequency domain. In addition, Tiwari et al. (2018) have shown that stock and CDS markets are net transmitters of volatility and degree of connectedness increases at higher frequency.

While previous literature has explored the relationship between stocks and bonds, it exhibits several limitations. Notably, much of the existing research does not examine extended time horizons, such as the one adopted in this study, which spans multiple crises. Furthermore, many empirical studies are constrained by the use of narrow and less comprehensive methodologies. These approaches often fail to account for the various quantiles of the distribution and their cross-predictability, instead focusing solely on the entire distribution or its upper extremes. Additionally, prior research has predominantly centred on returns, overlooking the valuable predictive insights that can be derived from volatilities. Finally, a significant portion of the literature emphasizes the U.S. market, often neglecting other influential markets, such as Germany.

## 2.2. Cross-Quantilogram

In economic and financial research, the issue of predictability is of paramount importance (Cowles and Jones, 1937). In line with this, numerous methodologies based on sign or rank statistics have been proposed, namely by Dufour et al. (1998), Christoffersen and Diebold (2002), among others.

One of such developments is the quantilogram approach introduced by Linton and Whang (2007), a tool for measuring predictability across different parts of the distribution of a stationary variable, based on the correlogram of "quantile hits". The authors applied a test for the hypothesis that a given time series exhibits no directional predictability. The null hypothesis posited by Linton and Whang (2007) asserts that the past information set of the stationary time series  $\{y_t\}$  does not enhance the prediction of whether  $\{y_t\}$  will fall below or above the unconditional quantile. Moreover, Han et al. (2016) argue that the quantilogram offers significant advantages over other test statistics for directional predictability, being conceptually appealing and easier to interpret. Additionally, the quantilogram approach is effective for heavy-tailed series, which are common in financial high-frequency data, as well as has the ability to consider very long lags, compared to regression-type methods, another notable advantage of the quantilogram.

This methodology has been extended by several authors, such as Davis and Mikosch (2009) who have introduced the extremogram, which applies the quantilogram to extreme quantiles. On the other hand, Davis et al. (2012) proposed statistical inference methods based on bootstrap and permutation for the extremogram. Further extensions include the work of Davis et al. (2013) who considered the quantilogram within a Fourier domain.

More recently, Han et al. (2016) proposed the cross-quantilogram to measure quantile dependence between two series, advocating statistical inference to test the hypothesis that one time series does not exhibit directional predictability over another. The authors established the asymptotic distribution of the cross-quantilogram and applied a stationary bootstrap procedure based on Politis and Romano (1994), to construct consistent confidence intervals. Furthermore, Pham (2021) highlighted the significant advantages of the cross-quantilogram, noting its robustness to misspecification errors and its flexibility in handling long lags compared to standard linear regression models. Additionally, the cross-quantilogram allows for the evaluation of directional transmission strength over different time periods without relying on movement conditions. The attributes of this

methodology enhance its robustness for analysing the relationship both between and within stocks and bonds, making it well-suited for the objectives of our study.

Empirical applications of the cross-quantilogram have predominantly focused on financial markets, particularly high-frequency data. Han et al. (2016) addressed stock return predictability, applying the cross-quantilogram to detect directional predictability from economic state variables to stock returns. The authors demonstrated that the cross-quantilogram provides a more comprehensive relationship between predictors and stock returns. Another application by Han et al. (2016) involved analyzing the systemic risk of individual financial institutions.

The cross-quantilogram has also found popular applications in ecological finance. Empirical research has explored topics such as decarbonization and green bonds. For example, Uddin et al. (2019) examined cross-quantilogram-based correlation and dependence between various asset classes and renewable energy stock returns, indicating a positive dependence of renewable energy stock returns on other asset returns, though this relationship did not hold when return series were in opposite quantiles. Pham (2021) evaluated the connectedness between green bonds and green equity across different investment horizons and market conditions, finding stronger connectedness during extreme market conditions. Razzaq et al. (2022) tested directional predictability between carbon trading and sectoral stocks in China using the cross-quantilogram. The study revealed an asymmetric dependence structure between carbon trading prices (CTP) and sectoral stocks, identifying negative directional predictability from CTP to stock market prices in bullish markets and positive predictability in bearish markets. Zhang et al. (2023) investigated whether the connectedness among fossil energy returns impacts renewable energy stock returns, concluding that only in extreme events such as the Global Financial Crisis, there is an interconnection between renewable and non-renewable energy markets.

The cross-quantilogram has also proven to be an effective statistical method for analyzing dependence between financial technology stocks and other asset classes. For instance, Karim et al. (2023) identified an asymmetric relationship between climate policy uncertainty and energy metals, using quantile causality and dependence analysis between financial technology stocks, green financial assets, and energy markets. Their study found that fintech prices were highly directionally predictable in all markets except green bonds in the lower quantiles of distributions, with negative predictability across all lag lengths in bullish states. Abakah et al. (2023) analyzed the distributional and directional predictabilities among fintech, bitcoin, and artificial intelligence stocks, finding that directional predictability among different assets was oscillatory over time lags, with stronger price connectedness for highly positive and negative changes.

Further applications of the cross-quantilogram have explored the relationship between stock market returns and other markets or indices. For instance, Kumar et al. (2021) utilized the cross-quantilogram to analyze the relationship between oil prices and stock market returns, investigating whether the inclusion of the geopolitical risk variable enhances the directional predictability from oil to stock returns. Their in-depth analysis differentiated between oil-exporting and oil-importing countries, revealing that the response of oil-exporting markets to energy shocks was higher and more persistent compared to oil-importing ones. Additionally, when the dependence structure of oil prices with stock markets was examined without accounting for geopolitical risk, no significant dependence was identified. However, upon conditioning for geopolitical risk factors, Kumar et al. (2021) found evidence of positive quantile dependence when both oil and stock returns were in the same quantiles of the distribution.

Lastly, Dai et al. (2022) analyzed cross-quantile dependence between the Chinese stock market, Chinese commodity market, crude oil, and investor sentiment. They identified a substantial degree of spillover among various Chinese commodity futures, a significant increase in total system spillover during major financial crises and the COVID-19 pandemic, and a high persistence of positive correlation between the stock index and investor sentiment index. Yet, Chang et al. (2024) employed data from the Taiwan Stock Exchange Weighted Index and the New Taiwan Dollar. Using the cross-quantilogram approach, they supported the flow-oriented hypothesis, demonstrating a negative Granger causality relationship from the New Taiwan Dollar to the Taiwan stock market, although this phenomenon persisted for only one day.

# 3. Methodology and Data

This section outlines the methodological approach implemented to perceive the relationship between bonds and stocks of the U. S. and Germany. Subsequently, we present the empirical strategy and the data.

## 3.1. Quantilogram setup

Assuming that  $(\mathbf{y}_t, \mathbf{x}_t, t \in Z)$  are a strictly stationary time series with  $\mathbf{y}_t = (y_{1t}, y_{2t})^T \in \mathbb{R}^2$  and  $\mathbf{x}_t = (x_{1t}, x_{2t}) \in \mathbb{R}^{d_1} \times \mathbb{R}^{d_2}$ , where  $x_{it} = \begin{bmatrix} x_{it}^{(1)} & \cdots & x_{it}^{(d_i)} \end{bmatrix}^T \in \mathbb{R}^{d_i}$  with  $d_i \in \mathbb{N}$  for i=1,2. Let  $F_{y_{i|x_i}}(\cdot | x_{it})$  denotes the conditional distribution function of the series  $y_{it}$  given  $x_{it}$  and  $f_{y_{i|x_i}}(\cdot | x_{it})$  denotes appropriate density function. Han et al. (2016) defined cross-quantilogram on the basis of the conditional quantile function defined as  $q_{i,t}(\tau_i) = \inf \left\{ v: F_{y_{i|x_i}}(v | x_{it}) \geq \tau_i \right\}$  for  $\tau_i \in (0,1)$ , for i=1,2.

The cross-quantilogram is a measure of serial dependence between two events  $\{y_{1t} \leq q_{1,t}(\tau_1)\}$  and  $\{y_{2t-k} \leq q_{2,t}(\tau_1)\}$  for any pair  $\tau = (\tau_1, \tau_2)^T \in \mathcal{T}$  and for an integer k, where  $\mathcal{T}$  is the range of quantiles someone is interested in evaluating directional predictability. Since  $\{y_{it} \leq q_{i,t}(\cdot)\}$  is called the quantile-hit or quantile-exceedance process (Linton and Whang, 2007), the cross-quantilogram is defined as the cross-correlation of the quantile hit processes:

$$\rho_{\tau}(k) = \frac{E\left[\psi_{\tau_1}\left(y_{1t} - q_{1,t}(\tau_1)\right)\psi_{\tau_2}\left(y_{2t-k} - q_{2,t-k}(\tau_2)\right)\right]}{\sqrt{E\left[\psi_{\tau_1}^2\left(y_{1t} - q_{1,t}(\tau_1)\right)\right]}\sqrt{E\left[\psi_{\tau_2}^2\left(y_{2t-k} - q_{2,t-k}(\tau_2)\right)\right]}},\tag{1}$$

for 
$$k = 0, \pm 1, \pm 2, ...$$
, where  $\psi_a(u) = 1[u < 0] - a$ .

In general, the cross-quantilogram is an extension of the quantilogram proposed by Linton and Whang (2007). While the quantilogram measures correlation between quantile-hit processes for a single time series, the cross-quantilogram captures serial dependence between the two series at different (or the same) conditional quantile levels.

The null hypothesis in the cross-quantilogram is:

$$H_0: \rho_\tau(1) = \rho_\tau(2) = \dots = \rho_\tau(k) = 0 \tag{2}$$

against the alternative hypothesis:

 $H_1: \rho_{\tau}(k) \neq 0$  for some k.

In order to test this hypothesis, Han et. al. (2016) proposed the following Ljung-Box statistic:

$$Q_{\tau}^{*}(p) = T(T+2) \sum_{k=1}^{p} \hat{\rho}_{\tau}^{2}(k) / (T-k),$$
(3)

and the sample cross-quantilogram  $\hat{\rho}_{\tau}^2(k)$  is calculated according to the following formula:

$$\hat{\rho}_{\tau}^{2}(k) = \frac{\sum_{t=k+1}^{T} \psi_{\tau_{1}}(y_{1t} - \hat{q}_{1,t}(\tau_{1})) \psi_{\tau_{2}}(y_{2t-k} - \hat{q}_{2,t}(\tau_{2}))}{\sqrt{\sum_{t=k+1}^{T} \psi_{\tau_{1}}^{2}(y_{1t} - \hat{q}_{1,t}(\tau_{1}))} \sqrt{\sum_{t=k+1}^{T} \psi_{\tau_{2}}^{2}(y_{2t-k} - \hat{q}_{2,t}(\tau_{2}))}}.$$
(4)

After calculating the values of the cross-quantilogram, the impact of financial stress index, monetary policy surprise shocks, fiscal stance and exchange rate are also studied. Hence, the values of the cross-quantilogram for sub-periods s=1,...,S are regressed on values of financial stress index, monetary policy shocks, fiscal primary balance and exchange rate. In addition, the vectors of the dependent and of the explanatory variable are defined for non-overlapping sub-periods of 18 months<sup>6</sup>. Specifically, we consider the following relationship:

$$CQ_s = f(FSI_s, MPS_s^{ECB}, MPS_s^{Fed}, CAPB_s^{DE}, CAPB_s^{US}, EXRATE_s^{DE\_US}, \varepsilon_s).$$
(5)

The vector  $EXRATE_s^{DE_US}$  consists of two variables:  $REER_s^{DE}$ , the real effective exchange rate for Germany and  $REER_s^{US}$ , the real effective exchange rate for the U. S., or another variable  $EXRATE_s^{USD/EUR}$ , i.e., the bilateral exchange rate between USD and EUR. *FSI* is the Financial Stress Index,  $MPS_s^{ECB}$  and  $MPS_s^{Fed}$  are the monetary policy surprise shocks of the European Central Bank (ECB) and the United States Federal

<sup>&</sup>lt;sup>6</sup> It should be stressed that the values of the cross-quantilogram measure presented in Tables 2-4, and A1-A3, in the Appendix, are calculated for overlapping windows. The choice of the 3-year rolling window enables an analysis of the cross-quantilogram measures for a sufficient number of observations and an analysis of the time-varying directional predictabilities for short periods. However, in the case of the regression model, the use of non-overlapping windows seems to be more appropriate. The longer is the sub-period (the cross-quantilogram is measured with higher precision), the shorter is the sample considered in the regression model explaining directional predictability in extreme quantiles. Using non-overlapping 18-months subperiods is a trade-off between the precision of estimation of the cross-quantilogram and the precision of estimation of a regression model.

Reserve (Fed), respectively.  $CAPB_s^{DE}$  and  $CAPB_s^{US}$  are the cyclically adjusted primary balances, CAPB, for Germany and the United States, respectively.

# 3.2. Empirical Strategy

To comprehensively analyze the relationship between stocks and bonds in the United States and Germany, we examine the rates of return, changes in yields, and volatilities across one, two, and three-day lags for both countries, given the weak seasonality effects on the financial markets (Flannery and Protopapadakis, 1988; Cho et al., 2007). Our analysis encompasses stock-stock and bond-bond interactions between the United States and Germany, as well as stock-bond interactions within each economy. To achieve robust results, it is imperative to select the most appropriate econometric models. The selection process requires identifying the order of integration of the variables included in the model. Although the cross-quantilogram values are computed based on stationary data, the time series of these values, which reflect causality performance over extended periods, may exhibit non-stationarity due to temporal changes. Similarly, exchange rates are expected to display non-stationarity. Hence, the cross-quantilogram values for stock market returns and changes in yields are calculated for various scenarios based on the following dimensions:

1. Between the U.S. and Germany stock markets, between U.S. and Germany bond markets, and between stock and bond markets within both countries, resulting in four variants;

2. For different daily lags, i.e., one, two, and three-day lags (1, 2, 3), totaling three variants;

3. In both directions, changing causes and effect results in two variants.

Ultimately, we then obtain a total of 24 time series (4\*3\*2) reflecting causality. However, we also consider the cross-quantilogram values for volatilities in both directions, between stock market volatilities, between bond market volatilities and both stock and bond market volatilities within both countries. Eventually, we obtain 8 (4\*2) more time series reflecting causality, so finally we have 32 time series. Naturally, stationarity may be present in some series, while absent in others. For the series where causality measures are integrated of order 1, we aim to identify long-run and short-run determinants using an error correction model. Conversely, for stationary causality measures, we estimate the parameters of standard linear regression with the first differences of explanatory variables integrated of order 1.

In the case of non-stationary causalities, we estimate the following Error Correction Model:

$$\Delta CQ_s = (\alpha_1 - 1) \left( CQ_{s-1} - x_{s-1}\hat{\lambda} \right) + z_s \mu + \varepsilon_s, \tag{6}$$

where the vector  $x_{s-1}$  consists of other I(1) variables,  $\hat{\lambda}$  is the vector of estimates of the cointegrating relationship and vector  $\mathbf{z}_s$  consists of stationary variables and first differences of I(1) variables.

An additional comment should be made regarding the strategy for handling low values of the cross-quantilogram that do not significantly differ from zero. We treat these values as non-zero for several reasons. The econometric methods suitable for identifying determinants of limited dependent and qualitative variables perform well with a large number of observations (see Maddala, 1983; Amemiya, 1985), which are typical in microeconomic analyses. In fact, models for limited dependent and qualitative variables, applied to time series generated by stochastic processes integrated of order 1, require an adequate number of observations (Grabowski and Welfe, 2016, 2020). Although we use daily data spanning from 1990 to mid-2024, the time series reflecting causality are calculated for non-overlapping windows. To obtain reliable results, these windows cannot be too narrow, which reduces their number. Furthermore, in the Tobit model (Tobin, 1958), we have an equation for the unobservable variable  $y_t^*$ , for which we know it is negative if the observable  $y_t$  equals zero. In our case, we have precise values of the crossquantilogram measure, even if they are very low, providing additional information compared to the Tobit model. This means our case differs from the traditional cases considered by Tobin (1958), where the observable expenditures for durable goods were zero.

# 3.3. Data

This study uses data for Germany and for the U. S. for the period between 02/01/1990 and 28/06/2024. Our key variables of interest are the stock returns calculated based on DAX 40 and S&P 500, the bond returns calculated on the 10-year German bond yields

and the 10-year U.S. bond yields, as well as the realized volatilities of both stocks and bonds. Data on these variables was retrieved from Thomson Reuters Datastream.

To analyze the impact of financial stress on time-varying causality among rates of return and volatilities we use data on the Financial Stress Index, which is available on FRED Database since 1990.<sup>7</sup> Once the values of the cross-quantilogram in the bad stance of financial markets are explained, measures of the Financial Stress Indicator should also reflect bad stance of financial markets. The higher value of the Financial Stress Index informs about stronger stress in financial markets. Therefore, in each case the quantile of order 0.95 is taken as an explanatory variable.

Data concerning the monetary policy surprises of the Fed and of the European Central Bank ( $MPS_s^{Fed}$  and  $MPS_s^{ECB}$ , respectively) are taken based on the research conducted by Jarociński and Karadi (2020). It should be stressed that data related to the monetary policy surprises of the European Central Bank have been available since 2000. Therefore, for the estimations in the second step, in that case, we considered that time span.

Lastly, for the second stage estimations, we included the cyclically adjusted primary balance of Germany and of the U.S. as percentage of GDP for the period between 1991 and 2023 ( $CAPB_s^{DE}$  and  $CAPB_s^{US}$ ), retrieved from the Bundesbank and the International Monetary Fund, respectively. Additionally, we have included the Real Effective Exchange rate of Germany and the U.S.as well as the Exchange rate of USD/EUR, collected from the BIS database between 1991 and 2023 ( $REER_s^{DE}$ ,  $REER_s^{US}$  and  $EXRATE_s^{USD/EUR}$ ).

In our empirical research, we analyze two distinct temporal dimensions. The first dimension utilizes daily data (five-day workweek) spanning from January 1990 to June 2024. The second dimension involves a time series observed at an 18-month frequency. Specifically, we have the following windows: the initial observation captures the performance of causality to other variables from January 1990 to June 1991, the second observation reflects the performance in the period between July 1991 and December 1992,..., and the last observation reflects the performance in the period between January 2023 and June 2024. The values of the causality measure, specifically the cross-

<sup>&</sup>lt;sup>7</sup> Typically, the Financial Stress Index is extracted from the website https://www.financialresearch.gov/financial\_stress\_index/. However, data is only available since 2000. Therefore, with the use of the relationship between Financial Stress Index and Kansas City Financial Stress Index, estimates of values of the FSI for years 1990-1999 are used.

quantilogram, are computed within these windows. Correspondingly, the other variables are adjusted to ensure that their values are recorded within the same 18-month intervals.

In the case of the monetary policy shocks, the variables  $MPS_s^{ECB}$  and  $MPS_s^{Fed}$ , their values are also aggregated in windows, since they take non-zero daily values at days of meetings devoted to changes of policy rate at the European Central Bank and the Fed.

Regarding the observations of for the  $CAPB_s^{DE}$  and for  $CAPB_s^{US}$  they were extracted on a yearly frequency and transformed into 18-monthly frequency, in a such way that the value for a given year is multiplied by (2/3) if the whole year constitutes a window and multiplied by (1/3) if half of a year constitutes a window. For example, if we calculate the  $CAPB_s^{DE}$  for a window between January 1990 and June 1991, we multiply the value of this variable from 1990 by (2/3), and for the values of this variable from 1991 we multiply them by (1/3). On the other hand, when we calculate the value of the  $CAPB_s^{US}$ for a window between July 1991 and December 1992, its value from 1991 is multiplied by (1/3) and its value from 1992 is multiplied by (2/3).

Lastly, in the case of exchange rate variables ( $REER_s^{DE}$ ,  $REER_s^{US}$  and  $EXRATE_s^{USD/EUR}$ ) monthly data from the end of months were downloaded. Therefore, the average values for 18-months are calculated for all windows and these values are used in the econometric model explaining time-varying causality.

In Figure 1, we present four graphical representations illustrating the evolution of Germany (blue) and the U.S. (red) Stock Indexes, 10-year bond yields, and the volatilities of stocks and bonds returns, from 1990 to 2024. The data clearly indicates that German stock indexes have experienced significantly higher growth compared to their U.S. counterparts, particularly in recent years. However, this growth is accompanied by notably higher return volatility in Germany over the period analyzed.

In terms of bond yields, the trajectories for both countries display a similar downward trend, closely mirroring each other. Despite this parallel movement, U.S. bond yields consistently exhibit slightly higher values relative to German bonds.



Notes: This figure displays four graphical representations of the evolution of Germany (blue) and the U.S. (red) Stock Indexes, Bond 10 years yields, and Stocks and bonds returns volatilities, between 1990 and 2024. Source: Author's own calculations.

Tab	ole	1.	Descri	ptive	statistics	of	the	daily	data.
								- /	

	Mean	Median	Std. Dev.	Min.	Max.	Obs.	Skewness	Kurtosis
Bonds US	-0.0004	-0.0018	0.0602	-0.5220	0.4370	8998	0,1541	6.1283
Bonds Germany	-0.0007	-0.0010	0.0466	-0.3880	0.3550	8998	-0,0146	6.8782
Stocks Returns US	0.0003	0.0007	0.0109	-0.0911	0.1014	8998	-0,2399	10.6329
Stocks Returns Germany	0.0003	0.0007	0.0139	-0.1038	0.1387	8998	-0,2521	9.4747
Volatility Stock Returns US	0.0011	0.0005	0.0019	0.0000	0.0341	8998	7,1779	79.5518
Volatility Stock Returns Germany	0.0017	0.0009	0.0029	0.0000	0.0379	8998	5,5867	48.3735
Volatility Bonds US	0.0327	0.0232	0.0318	0.0012	0.3982	8998	3,5771	24.2267
Volatility Bonds Germany	0.0196	0.0124	0.0225	0.0000	0.2685	8998	4,1657	31.5184
FSI <sub>s</sub>	0.6516	0.2412	1.3420	-0.6905	5.6964	23	2.3333	9.6913
$MPS_{s}^{Fed}$	-0.1413	-0.0798	0.3237	-0.9108	0.2716	23	-1.2187	3.6631
MPS <sup>ECB</sup>	0.0424	0.0589	0.1604	-0.2890	0,4227	17	0,1165	4.0001
$CAPB_{s}^{US}$	-2.3697	-2.5226	2,4401	-5,4150	1,3029	23	0,2153	1.4322
$CAPB_{S}^{DE}$	0.2300	0.0794	0.9496	-1.9181	1.6858	23	-0.1495	2.4682
REER <sup>US</sup>	87.2131	84.8333	9.1338	76.5444	104.1456	23	0.5059	1.8495
$REER_s^{DE}$	102.5884	101.0939	3.9906	96.4817	113.2556	23	0.9981	3.5741
$EXRATE_{s}^{DE_{US}}$	0.8524	0.8324	0.0976	0.7045	1.1176	23	0.8357	3.7058

Notes: This table displays a summary statistic (Mean, Median, Standard Deviation, Minimum, Maximum, number of observations, Skewness, and kurtosis of all variables employed in this study for the period of 1990 to 2024. Source: Author's own calculations.

Table 1 provides the summary statistics for all variables used in this study. Notably, the table highlights the negative bond returns observed, as well as the similarities between German and U.S. stock returns. However, it is evident that German stock returns and U.S.

bond returns exhibit high volatility. Additionally, the presence of more extreme deviations from the mean are reported for U.S. stock returns and German bond return volatilities.

# 4. Empirical Analysis

# 4.1. Baseline Results

# 4.1.1. Returns

Initially, we investigate the relationships between stock market returns and fluctuations in 10-year sovereign bond yields for the United States and Germany, across various quantiles of the distribution, using data from 1990 to mid-2024.

Figure 2 illustrates the causality relationship between U.S. and German stock returns quantiles with a one-day lag. The left chart of Figure 2, which depicts U.S. stock returns causing German stock returns across different quantiles, highlights a positive and stronger causality, particularly when the analysis focuses on the highest quantiles of stock returns. Simultaneously, the lowest returns of U.S. stock markets exhibit a stronger causality, in the same direction, with German stock returns caused by German stock markets (Figure 2, right chart). Specifically, lower returns, depicted by the lowest quantiles, correspond to lower returns in the other market on the following day.

While the negative returns between markets are more synchronized in their causality (as seen in both charts of Figure 3 for the lowest stock returns quantiles), the relationship in the highest stock returns quantiles shows less synchronization between the two markets. Additionally, a pattern emerges relating opposite extreme quantiles of stock returns between the United States and Germany. Specifically, the lowest (highest) stock return quantiles in the U.S. tend to cause higher (lower) returns in German stock markets. The same rationale is observed when the German market causes the stock returns in the U.S. stock market on the following day, consistent with the findings of Bonfiglioli and Favero (2005) on the co-movements between U.S. and German stock markets.

Figure 4 demonstrates that the bidirectional predictability between the quantiles of German bonds and U.S. bonds exhibits a similar relationship to that observed in Figure 3. However, across quantiles, the degree of causality is more uniform from the United States to Germany. Notably, for the top quantiles of bond returns from the source market (United States in the left chart, Germany in the right chart), there appears to be a reduction

in returns in the following day in the sovereign debt market of the other country. This behavior may indicate a flight to safety, as higher bond returns correspond to lower bond prices, signaling increased risk. Consequently, financial investors may deleverage their positions and allocate investments to safer markets, whose behavior is consistent with the nonlinearity and flight-to-safety in the risk-return trade-off for stocks and bonds (Adrian et al., 2019).

When examining the cross-relationship between stocks and bonds, we observe that the strongest causality persists at the extremes of the quantiles for Germany and the United States, as depicted in Figures 4 and 5, respectively. However, these causal relationships are weaker compared to those observed between stocks-stocks and bonds-bonds. Furthermore, Figures 4 and 5 illustrate that the cross-quantile relationship from bonds to stocks is more heterogeneous, whereas the causality from stocks to bonds is more homogeneous and approaches a non-significant relationship (with causalities varying between -0.05 and 0.05). Even within the stock-to-bond causality relationships, the United States emerges as a more dynamic case of stock-to-bond causality. These observations are consistent with the findings of Baur and Lucey (2009), who analyzed flights to quality and contagion in stock-bond correlations, thereby providing further support for the observed patterns.



#### Figure 2. Values of cross-quantilogram indicating directional predictability between stock markets with 1-day lag.

Source: Authors' calculations.





From the United States to Germany

■-0,05-0 ■0-0,05 ■0,05-0,1 ■0,1-0,15

From Germany to the United States



■-0,05-0 ■0-0,05 ■0,05-0,1 ■0,1-0,15

Source: Authors' calculations.



#### Figure 4. Values of cross-quantilogram indicating directional predictability between stock and bond market for Germany (1-day lag).

Source: Authors' calculations.





■-0,1--0,05 ■-0,05-0 ■0-0,05 ■0,05-0,1



From Stocks to Bonds

■-0,1--0,05 ■-0,05-0 ■0-0,05 ■0,05-0,1

Source: Authors' calculations.

# 4.1.2. Volatilities

In this subsection, we examine the predictability of volatilities in the stock and bond markets of Germany and the U.S. From Figures 6 and 7, we can see that causality between the S&P 500 and the DAX 40 stock and bond returns volatilities is not uniformly distributed across quantiles. At lower quantiles (0.0 to 0.01), the causality measures are minimal, indicating little influence of low S&P 500 volatilities on DAX 40 volatilities, and vice versa. However, as the quantiles increase, there is a noticeable rise in causality measures, illustrating that cross-country predictability is more pronounced in the upper quantiles of both stocks and bonds, exhibiting a smooth inclined surface across the quantiles. Additionally, cross-country predictability in stocks (from the U.S. to Germany and vice versa) shows the highest coefficients, ranging between 0.4 and 0.5, indicating a strong predictive relationship.

Regarding cross-asset predictability, Figures 8 and 9 also reveal an upper slope surface towards the upper quantiles, although it is less smooth and smaller compared to Figures 6 and 7. Specifically, certain mid-range quantiles of the German bond yields (around 0.50 to 0.70) in Figure 8 exhibit sporadic spikes in causality measures, affecting specific quantiles of the DAX 40 stock returns. These irregularities suggest that there are complex dynamics at play, possibly influenced by market-specific events or broader economic factors that affect both bond and stock markets differently at various volatility levels.

In the case of the U.S., we can observe in Figure 9 that the unidirectional relationship between bonds to stocks presents a similar pattern throughout all quantiles of the distribution, with a relative spike in the highest quantiles.

In conclusion, the causality analysis between the volatilities of changes in yields of United States and Germany sovereign bonds and the volatilities of S&P 500 and DAX40 stock returns demonstrates a clear dependency, particularly at higher volatility levels. The findings underscore the importance of considering volatility quantiles when assessing market interdependencies, as the impact is not uniform across different volatility levels. Specifically, for highly turbulent periods in financial markets, when typically, financial assets vary the most, they show some degree of mutual influence. This nuanced understanding can aid in better risk management and forecasting in financial markets, providing valuable insights for investors and policymakers.



#### Figure 6. Values of cross-quantilogram indicating directional predictability between stock market volatilities with 9-days lag.

Source: Authors' calculations.





■0-0,1 ■0,1-0,2 ■0,2-0,3

From Germany to the United States



■-0,1-0 ■0-0,1 ■0,1-0,2 ■0,2-0,3 ■0,3-0,4

Source: Authors' calculations.



#### Figure 8. Values of cross-quantilogram indicating directional predictability between stock and bond market volatilities for Germany (9-days lag).

Source: Authors' calculations.



#### Figure 9. Values of cross-quantilogram indicating directional predictability between stock and bond market volatilities for the U.S. (1-day lag).

■-0,1-0 ■0-0,1 ■0,1-0,2 ■0,2-0,3





■0-0,05 ■0,05-0,1 ■0,1-0,15 ■0,15-0,2 ■0,2-0,25

Source: Authors' calculations.

# 4.2. Results on extreme quantiles

In this section, we investigate the dynamic predictability between the financial markets of Germany and the United States, with a particular emphasis on stock and bond markets and their respective volatilities, utilizing three-year rolling windows. Specifically, we analyze directional predictability across various lags (ranging from one to three periods) and different sub-periods, assessing the predominant direction of causality, i.e., from Germany to the United States or vice versa, when analyzing the same asset (bonds or stocks) or within countries, when we intend to analyze the direction of the predictability from stocks to bonds and vice versa. These analyses encompass both rates of return (yield changes) and the volatilities of stock and bond markets.

To account for the time dimension and analyze how causalities evolved, we focus on specific quantiles. We concentrate on extreme quantiles that reflect negative performance in stock markets and negative performance in bond markets (from the perspective of debt servicing costs). We analyze time-varying causality for the quantile of order 0.05 in stock markets and for quantile of order 0.95 in bond markets. When examining causality in volatilities, we focus on the 0.95 quantile, as higher volatility levels are associated with a more difficult stance of financial markets.

Lastly, we emphasize that for all cases, 10,000 bootstrap replications are performed to calculate critical values. Due to the use of bootstrapped critical values, the significance is not always an increasing function of the cross-quantilogram estimate. In some instances, a cross-quantilogram with a lower estimate may exhibit higher significance than one with a higher estimate.

# 4.2.1. Returns and changes in yields

In Table 2, we present the estimated values of bidirectional predictability between stock returns in the United States and Germany, specifically focusing on the 0.05 quantile, which represents the lowest 5% of stock returns. These estimates are calculated over three-year rolling windows, providing insights into the extreme negative tail dependencies and the dynamic interactions between the two markets.

The period from 1991 to 1993 exhibits a significant positive relationship between U.S. and German stock returns at a two-day lag  $(0.157^{***})$ , indicating that extreme negative returns in the U.S. market are followed by similar movements in the German market after two days. This period aligns with the aftermath of the reunification of Germany and the Gulf War, both of which had substantial global economic implications. Additionally, a significant negative

causality is observed from German to U.S. stock returns at a three-day lag (-0.053\*\*\*), suggesting that extreme negative returns in the German market are followed by opposite movements in the U.S. market after three days. This phenomenon can be attributed to the differing economic conditions and recovery trajectories of the two countries during the early 1990s. The Gulf War, which began in August 1990 and ended in February 1991, had indeed a notable impact on stock returns, particularly in the short term. When the conflict broke out, there was a significant initial decline in stock markets due to heightened uncertainty, registering a decline of the S&P 500 by 20%, approximately. The results of the contagion effect between stocks are consistent with the findings of Bonfiglioli and Favero (2005).

The period from 2006 to 2008 is marked by multiple significant positive dependencies, indicative of heightened interconnectedness during the global financial crisis. The bidirectional causality in stock returns between the United States and Germany exhibits significant positive dependencies at various lags, with the most pronounced effect observed at a three-day lag. The global financial crisis, precipitated by the collapse of Lehman Brothers in 2008, resulted in severe market turmoil and heightened volatility. The significant positive dependencies identified during this period underscore the contagion effects and the transmission of financial shocks across international borders, as extensively documented in the literature on financial contagion (Longin and Solnik, 2002; Forbes and Rigobon, 2002; Grammatikos and Vermeulen, 2012; Bekaert et al., 2014).

The period from 2018 to 2020 also exhibits strong bidirectional spillovers, with significant values for both directions, i.e., fromU.S.to DE, DE to US. The causal relationship fromU.S.to Germany is notably strong at a one-day lag (0.236\*\*\*), suggesting immediate spillover effects from the U.S. to the German market. This period includes significant geopolitical events such as the US-China trade war, notably with the introduction of tariffs and trade barriers (see Afonso et al., 2024b) and Brexit, which created uncertainty and volatility in global markets. The significant positive dependencies during this period align with the findings of previous studies on the impact of geopolitical risks on stock market correlations (Bekaert et al., 2014).

Additionally, we examine the cross-quantilogram relationship of bond market returns between Germany and the United States. The significant relationships identified in the early 1990s, particularly during the 1992-1994 period, where the U.S. to Germany lagged effects at one-day and two-days are positive and significant at the 5% and 1% levels, respectively, can be attributed to the economic adjustments following the reunification of Germany. The reunification imposed substantial fiscal and economic challenges, rendering the German bond

market highly sensitive to external shocks, especially from the U.S. These results are consistent with those for stocks.

The late 1990s and early 2000s, characterized by significant cross-quantilogram values, coincide with the dot-com bubble and its subsequent burst. The period 1998-2000, for instance, shows a significant influence of U.S. bond market returns on German returns at the one-day lag (0.085\*\*\*) and three-days lag (-0.052\*\*\*). The dot-com bubble was a period of excessive speculation in technology stocks, leading to a market crash that had global repercussions.

The early 2000s also saw the implementation of the Economic and Monetary Union (EMU) and the introduction of the euro, which had a significant effect on bond markets. The convergence of bond yields across Eurozone countries, notably towards the German yields, was influenced by the Stability and Growth Pact and the perceived fiscal discipline among euro area member states. However, the early 2010s brought the European sovereign debt crisis, where fears of fiscal unsustainability in peripheral Eurozone countries led to significant yield spikes and contagion effects (Afonso et al., 2024a). The significant cross-quantilogram values during this period underscore the vulnerability of Germany to the influence of events coming from the U.S., as highlighted by Lakdawala et al. (2021). Furthermore, the results underscore the potential implications of fiscal unsustainability. Higher changes in yields, as captured by the top quantile, often signal concerns about fiscal health and the risk of default. The significant cross-quantilogram values, particularly those indicating positive dependence, suggest that fiscal instability in one country can quickly influence investor perceptions and yield movements in another. This is particularly relevant in the context of the European sovereign debt crisis, where fears of lacking fiscal sustainability led to significant yield spikes and contagion effects across global bond markets.

	R	elationship b	etween stock	market return	s of order 0.0	05	Relation	nship between	changes in s	overeign bon	d yields of or	der 0.95
Window	Lag	<u>z=1</u>	Lag	g=2	Las	g=3	La	g=1	La	g=2	Las	g=3
	$US \rightarrow DE$	$DE \rightarrow US$										
1991-1993	-0.001	0.078	0.157***	-0.028	0.104**	-0.053***	-0.019	-0.019	0.040	0.010	0.069	0.010
1992-1994	-0.001	0.025	-0.028	-0.027	-0.001	-0.053***	0.095**	0.124	0.153***	0.012	0.066	0.071
1993-1995	-0.001	0.078*	-0.028**	0.027	-0.028***	-0.053***	0.059	0.115	0.115*	0.004	0.087*	0.087**
1994-1996	-0.054***	0.025	-0.001	-0.028***	-0.028***	-0.054***	0.078*	0.130*	0.025	-0.028***	-0.001	-0.001
1995-1997	-0.001	0.078	0.104*	-0.028	0.051	0.000	0.052	0.078*	-0.027***	0.000	0.000	0.000
1996-1998	0.157***	0.157**	0.104*	-0.001	-0.001	0.051	-0.001	0.104	-0.053***	-0.001	0.051*	-0.054***
1997-1999	0.104*	0.104*	0.104**	-0.001	-0.001	-0.028**	0.080	0.107*	0.001	0.000	-0.053***	-0.053***
1998-2000	0.078	0.051	0.080	-0.001	-0.027	-0.028***	0.085	0.167***	0.003	0.003	-0.025	-0.052***
1999-2001	0.078	0.025	0.078	0.025	0.025	-0.054***	0.107**	0.160**	-0.027***	0.027	0.000	0.028
2000-2002	0.078	0.107	0.104	-0.026	0.025	0.032	0.001	0.136*	0.028	0.136*	0.001	0.028
2001-2003	0.104	0.104	0.104	-0.001	0.078	0.027	0.001	0.217***	0.001	0.136***	0.001	0.028
2002-2004	0.157*	0.131**	0.104	0.025	0.051	0.051	-0.025	0.217***	0.001	0.082	0.028	0.001
2003-2005	0.107**	0.052	0.080	0.104*	0.053	0.051	-0.024	0.194**	-0.023	0.004	0.087*	0.062
2004-2006	0.130**	0.051*	0.000	0.051	0.080	0.025	0.115*	-0.023	-0.024	-0.051***	0.059	-0.051***
2005-2007	-0.001	-0.001	0.025	0.025	0.025	0.104**	0.112*	0.112*	0.003	-0.025	0.085***	0.004
2006-2008	0.157*	0.131	0.130*	0.078	0.157**	0.209***	0.136***	0.109**	0.082*	0.082	0.055	0.055
2007-2009	0.131**	0.157***	0.104*	0.052	0.157**	0.130**	0.104*	-0.001	0.025	0.025	0.051	0.025
2008-2010	0.131**	0.131**	0.131**	0.025	0.157***	0.104**	0.083	-0.025	0.055	0.001	0.085**	0.028
2009-2011	-0.001	0.078	0.051	-0.001	0.025	0.025	0.130**	-0.054***	0.078*	-0.027***	-0.001	0.001
2010-2012	0.025	0.104**	0.078	0.051	0.051	0.051	0.080	0.027	0.107**	-0.027***	0.000	-0.027***
2011-2013	0.025	0.078	0.130*	0.078	0.104*	0.104**	0.163***	0.001	0.082*	0.028	-0.053***	-0.053***
2012-2014	0.025	0.052	0.104	0.000	0.051	0.027	0.109*	0.028	0.055	0.028	0.001	-0.026
2013-2015	0.025	0.131**	0.078	0.025	0.025	0.051	0.052	0.000	0.053	0.03	-0.053***	0.003
2014-2016	0.078	0.104	0.025	0.025	-0.001	0.051	0.025	0.025	-0.001	-0.001	-0.054***	0.025
2015-2017	0.078	0.130**	0.025	0.051	-0.054***	0.157***	0.025	0.025	0.000	-0.001	-0.053***	-0.001
2016-2018	0.130*	0.078	0.053	0.051	0.000	0.130**	0.107**	0.025	0.000	-0.028***	-0.027***	0.078*
2017-2019	0.104**	0.104*	-0.001	0.051	-0.001	0.080	-0.027***	0.053	0.080	0.000	0.028	0.055
2018-2020	0.236***	0.183**	0.052	0.157**	0.051	0.104	0.054	0.134**	0.133**	0.107**	0.107	0.133***
2019-2021	0.210***	0.157	0.052	0.157**	0.078	0.078	0.025	0.078	0.131*	0.078	0.051	0.104*
2020-2022	0.157**	0.157**	0.051	0.157**	0.051	0.051	0.107*	0.027	0.053	0.027	0.082	0.055
2021-2023	0.104*	0.051	0.025	0.051	0.051	0.027	0.085	-0.052***	0.085*	0.003	0.030	-0.025
2022-2024Q2	0.106	0.074	0.042	0.042	0.074	0.042	0.012	-0.053***	0.044	-0.020	-0.020	-0.053***

Table 2. Cross-quantilogram values for 3 years-rolling-windows reflecting relationships between stock market returns in Germany and the United States in quantiles of order 0.05, and relationships between bond market returns in Germany and the United States in quantiles of order 0.95.

We turn our attention now to the cross-asset predictability in both Germany and the U.S.. The analysis of the cross-quantilogram results for the three-year rolling windows in Germany, reflecting the relationship between the lowest 5% stock returns (quantile of order 0.05) and the top 5% changes in bond yields (quantile of order 0.95), presented in Table 3, provides significant insights into the relationship between extreme movements in the stock and bond markets. Within the German market, the bidirectional relationship between bonds and stocks was predominantly positive and significant until the early 21<sup>st</sup> century, coinciding with the start of the Economic and Monetary Union. This predictable relationship was observed across all lags. Interestingly, the directional relationship from stocks to bonds became more significant after the 21<sup>st</sup> century, with the sign varying over the years. Specifically, during the sovereign debt crisis, changes in stocks led to opposite changes in bonds. This shift could be attributed to increased market volatility and investor behavior during periods of economic uncertainty, where investors often move their capital between stocks and bonds to manage risk and seek safer investments.

Furthermore, we analyze the cross-relationship between the stock and bond markets for the United States. We highlight the positive bilateral predictability between stock and bond returns in the early 2000s, indicating mutual influences between the two markets. This period coincides with the implementation of the Gramm-Leach-Bliley Act, which significantly impacted financial markets by repealing parts of the Glass-Steagall Act and allowing commercial banks, investment banks, and insurance companies to consolidate. This deregulation contributed to increased market volatility and interdependence between stock and bond markets. Additionally, the 2010s also exhibit periods of positive and significant cross-asset predictability. This period was characterized by uncertainty due to fluctuations in commodity prices, variations in Federal Reserve interest rates, and geopolitical tensions with China. According to Ferrer et al. (2019), during turbulent times, stocks and bonds tend to exhibit positive co-movement.

	Germany United States											
Window	Lag	g=1	Lag	g=2	Lag	g=3	Lag	g=1	La	g=2	Lag	g=3
	$B \rightarrow S$	$S \rightarrow B$	$B \rightarrow S$	$S \rightarrow B$	$B \rightarrow S$	$S \rightarrow B$	$B \rightarrow S$	$S \rightarrow B$	$B \rightarrow S$	$S \rightarrow B$	$B \rightarrow S$	$S \rightarrow B$
1991-1993	-0.052	-0.052	0.028**	0.054***	-0.078*	-0.027	-0.098*	0.019	-0.010	-0.040	-0.042	0.019
1992-1994	-0.114	0.028**	0.027	0.001	0.027*	-0.025	0.020	0.020	-0.039	-0.040	-0.039	-0.010
1993-1995	-0.163**	0.026	0.026	-0.028	-0.001	-0.001	-0.028	0.026	-0.057	-0.028	-0.003	-0.109**
1994-1996	-0.183***	0.054***	0.001	0.001	0.028***	0.001	-0.051	0.001	0.001	-0.078*	0.001	-0.078
1995-1997	-0.025	0.028**	0.027	0.053***	0.027*	-0.028	-0.078*	0.001	0.028***	0.028***	-0.025	0.028***
1996-1998	0.027**	0.001	0.028*	0.054***	0.028***	0.001	-0.025	0.027**	0.028**	0.001	0.001	0.028***
1997-1999	0.054***	-0.025	0.028***	0.027	-0.025	0.000	-0.053*	0.000	0.027	-0.027	-0.053	0.000
1998-2000	-0.001	-0.001	0.025	-0.001	-0.057	-0.001	0.000	0.027	0.027**	0.000	-0.053	0.000
1999-2001	-0.025	0.028***	0.001	-0.051	-0.080*	0.001	-0.053*	-0.027	-0.053	0.027	0.000	0.027**
2000-2002	0.053***	0.025	0.027	0.024	0.000	0.050***	0.028***	-0.027	-0.078**	0.027*	-0.027	-0.053
2001-2003	0.054***	-0.025	0.001	-0.051	0.001	0.028***	0.026	-0.001	-0.001	0.026	0.025	-0.055
2002-2004	0.053***	-0.027	0.000	-0.053	0.027	0.027	0.053***	-0.054	0.027	-0.053	0.000	-0.160***
2003-2005	0.023	-0.058	-0.004	-0.06	0.024	0.051***	0.054***	0.028*	0.028***	0.027	-0.078*	-0.001
2004-2006	-0.087	-0.004	0.022	-0.004	-0.062	-0.032	0.028***	0.028**	0.028***	0.028***	0.028***	0.027**
2005-2007	-0.08	-0.027	-0.053	-0.027	-0.027	-0.027	0.026	0.026	-0.055	-0.028	0.053***	-0.085*
2006-2008	-0.055	-0.163**	-0.028	-0.055	-0.001	-0.109**	-0.052	-0.210**	-0.078	-0.157**	0.001	-0.104*
2007-2009	-0.025	-0.104*	-0.025	-0.052	0.001	-0.104	0.001	-0.183***	-0.078	-0.131**	-0.051	-0.051
2008-2010	0.026	-0.054	0.000	-0.053	0.000	-0.107	0.026	-0.160***	-0.053	-0.133***	-0.055	-0.053
2009-2011	-0.051	-0.078*	-0.025	0.028***	-0.051	-0.130***	0.001	-0.078	-0.078*	-0.133**	-0.078	-0.028
2010-2012	-0.053	-0.080*	-0.027	0.027**	-0.053	-0.133***	0.054***	-0.078	0.001	-0.078	0.001	-0.104*
2011-2013	-0.082	-0.055	-0.028	0.026	-0.055	-0.055	0.028***	-0.104	0.028***	-0.078	0.001	-0.104*
2012-2014	-0.001	-0.028	0.026	0.026	0.026	-0.028	0.001	0.028**	0.000	-0.051	-0.027	-0.078
2013-2015	0.027	0.001	-0.028	-0.080**	-0.001	-0.080**	0.001	0.001	0.001	0.000	0.001	0.027*
2014-2016	-0.025	-0.025	0.001	-0.025	-0.051	-0.025	0.028**	0.028**	0.028***	0.028***	0.028***	0.028***
2015-2017	0.001	-0.025	-0.051	-0.053	-0.051	-0.053	0.054***	-0.025	0.001	-0.025	0.028***	0.001
2016-2018	-0.025	-0.027	0.027*	0.027*	-0.027	0.053***	-0.025	0.054***	-0.025	0.028***	0.054***	0.028***
2017-2019	-0.027	0.053***	0.027*	0.027	0.026	0.052***	0.001	0.054***	0.001	0.028***	0.054***	0.001
2018-2020	-0.054	-0.054	0.027	-0.053	-0.027	-0.027	-0.025	-0.104	-0.078	-0.078	-0.025	-0.104*
2019-2021	-0.052	-0.052	0.001	-0.078	-0.025	-0.025	-0.052	-0.104*	-0.052	-0.052	-0.025	-0.104*
2020-2022	-0.052	-0.052	0.001	-0.025	0.000	-0.051	-0.027	-0.107**	-0.053	-0.08	-0.136**	-0.107**
2021-2023	-0.027	-0.053	-0.053	-0.027	-0.080	-0.027	-0.028	-0.082	0.026	-0.055	-0.003	-0.055*
2022-2024Q2	-0.012	-0.044	-0.044	-0.044	-0.109	-0.044	-0.010	-0.074	0.022	-0.042	0.022	-0.042

 Table 3. Values of cross-quantilogram for 3 years-rolling-windows reflecting relationship between stock (quantile of order 0.05) and bond market (quantile of order 0.95) returns in Germany and the United States.

# 4.2.2. Volatilities

Having analyzed the cross-country relationships for stock returns and changes in sovereign bond yields, as well as the causality between stocks and bonds for Germany and the United States, in this subsection we rely on the volatilities of abovementioned relationships.

The analysis of stock returns and changes in sovereign yields is incomplete without a thorough examination of their volatilities. In the realm of portfolio management, volatility analysis is indispensable for constructing diversified portfolios. By evaluating the volatility of various assets, namely stocks and sovereign bonds, investors can optimize the risk-return profile of their portfolios, thereby enhancing their investment strategies. Furthermore, volatility serves as a barometer of market sentiment and investor behavior. Additionally, changes in sovereign yields and their volatility can provide valuable insights into shifts in economic conditions, monetary policy, and investor confidence, making them crucial indicators for macroeconomic analysis and forecasting.

To analyze the causality for all the previously described relationships, we focus on the volatilities over non-overlapping 9-day intervals (t - 4, t, and t + 4) of stock returns and changes in sovereign yields. The results are detailed in Table 4.

During the early 1990s, specifically from 1992 to 1996, the causality from U.S. stock returns volatility to DE stock returns volatility, and vice versa, is consistently negative and statistically significant at the 1% level. However, this effect is also present in the bond markets transmission from Germany to the U.S. and within the domestic market of the U.S. (effect that prolongs until the 2000s). This period coincides with the early 1990s recession, characterized by restrictive monetary policies, the 1990 oil price shock, and the savings and loan crisis in the United States.

As expected, the financial crisis of 2007-2008 had a profound impact on the volatility dynamics between U.S. and DE stock and bond returns as markets move in the same direction, however, in the 2010's the causality is predominantly negative. Align with the findings of Diebold and Yilmaz (2012) and Adrian et al. (2019) regarding the flight-to-safety effect, the results for the 2006-2008, 2007-2009, and 2008-2010 periods show significantly positive causality at the 1% level. These findings support dynamic asset pricing models that suggest the risk premium varies nonlinearly with market volatility. In fact, this period was characterized by severe market turmoil, with the collapse of major financial institutions and unprecedented levels of market volatility. We argue that this synchronized movement is driven by fear and uncertainty, leading to a broad sell-off in riskier markets and a surge in safer ones. After the initial panic subsides, markets start to move in opposite directions as investors reassess the situation and look for opportunities. Different assets

respond differently to new information and changing economic conditions. For example, stocks might recover as confidence returns, while bonds might decline if interest rates rise.

In the late 2010's period, the relationship between and within U.S. and DE returns volatility remains significant and positive. The 2018-2020 and 2019-2021 periods show significantly positive causality at the 1% level, suggesting that the interconnectedness between these markets persisted even during the recovery phase. This finding aligns with the increased globalization of financial markets and the continued influence of U.S. market movements on global financial stability.

<b>TT</b> 7• 7	Stock v	s. Stock	Bond v	s. Bond	DE (betwee	en markets)	US (betwee	en markets)
Window	$US \rightarrow DE$	$DE \rightarrow US$	$US \rightarrow DE$	$DE \rightarrow US$	$B \rightarrow S$	$S \rightarrow B$	$B \rightarrow S$	$S \rightarrow B$
1991-1993	0.106	-0.002	0.055	-0.054***	0.160	-0.026	-0.028***	-0.054***
1992-1994	-0.054***	-0.052***	0.262**	0.057	-0.025	0.025	-0.052***	0.077
1993-1995	-0.053***	0.055	0.217**	-0.053***	-0.026	0.028	-0.053***	0.163**
1994-1996	-0.052***	-0.055***	0.051	-0.055***	0.002	0.025	0.104	-0.055***
1995-1997	0.089	0.064	-0.050***	-0.054***	-0.054***	-0.044***	-0.028***	-0.051***
1996-1998	0.262*	0.394***	-0.002	-0.054***	-0.002	0.077	0.025	0.183
1997-1999	0.183	0.183	0.077	-0.054***	0.130	0.077	0.051	0.288**
1998-2000	0.376***	0.183	0.106	-0.028***	0.202	0.183*	0.080	0.266*
1999-2001	0.230**	0.244**	0.104	-0.027***	-0.023	-0.053***	-0.028***	-0.054***
2000-2002	0.183	0.142	0.136	0.104	-0.028***	-0.022	0.077	-0.028***
2001-2003	0.156	0.025	-0.054***	0.008	-0.054***	-0.054***	0.130	0.037
2002-2004	0.446***	0.130	-0.054***	0.104*	-0.028***	0.025	0.157	0.130*
2003-2005	0.441***	0.355**	-0.054***	0.104	0.326**	0.104	-0.05***	-0.054***
2004-2006	-0.002	0.183	-0.055***	0.051	0.288***	0.051	0.077	0.025
2005-2007	-0.053***	0.190	0.284	0.136	0.190*	-0.053***	0.103	0.136
2006-2008	0.526***	0.446***	0.480***	0.183	0.315	0.209	0.240	0.236
2007-2009	0.526***	0.446***	0.315***	0.077	0.288	0.130	0.130	0.183
2008-2010	0.526***	0.446***	0.236**	-0.002	0.288	0.077	0.104	0.130
2009-2011	-0.002	0.051	0.051	-0.021	0.156	0.156	0.130	0.152
2010-2012	0.130	0.394***	0.077	-0.028***	0.104	0.104	0.183*	0.025
2011-2013	0.156	0.499***	0.130	-0.002	0.130	0.104	0.315***	0.025
2012-2014	0.053	-0.028***	-0.049***	-0.054***	-0.054***	0.341***	-0.049***	0.080
2013-2015	0.051	0.307***	-0.050***	-0.052***	-0.025	0.07	-0.022	-0.028***
2014-2016	-0.054***	0.156	-0.054***	-0.054***	0.025	0.025	-0.028***	-0.054***
2015-2017	-0.055***	0.156	-0.055***	-0.027***	0.025	0.025	-0.028***	-0.054***
2016-2018	-0.042***	0.080	-0.05***	-0.050***	0.143	-0.05***	-0.027***	-0.048***
2017-2019	0.156	0.130	-0.028***	-0.050***	0.124	-0.055***	0.025	-0.055***
2018-2020	0.631***	0.526***	0.262	0.183	0.394***	0.367***	0.526***	0.394***
2019-2021	0.631***	0.526***	0.487***	0.209	0.552***	0.595***	0.526***	0.394**
2020-2022	0.420**	0.499***	0.051	0.025	0.025	-0.054***	0.526***	0.341**
2021-2023	0.163	0.109	0.109	-0.026	-0.026	0.136	0.298**	-0.053***
2022-2024Q2	0.201	0.105	0.105	-0.023	-0.055***	0.105	0.265**	-0.055***

Table 4. Values of cross-quantilogram for 3 years-rolling-windows reflecting relationship between volatilities in stock markets for quantiles of order 0.95. (Lag=9)

# 4.3. The impact of Financial Stress Index and Monetary Policy Shocks

The calculations presented in the preceding sections were designed to elucidate the dependency of causality between stock market returns, changes in yields, and volatilities on the quantiles of their respective distributions (Figures 1-8). Furthermore, these calculations highlighted the temporal fluctuations of the causality measure within three-year overlapping windows. In this section, our objective is to discern the influence of various categories on these time-varying causality measures.

The causality measures calculated in the previous subsection are presented within overlapping windows. However, to obtain a time series of independent observations, it is necessary to calculate these values within non-overlapping windows. Consequently, we generated time series for 48 variants, with causality measures computed in non-overlapping windows.

Subsequently, we employed standard unit root tests, specifically the Augmented Dickey-Fuller (ADF) test (Dickey and Fuller, 1979, 1981) and the Kwiatkowski-Phillips-Schmidt-Shin (KPSS) test (Kwiatkowski et al., 1992), to determine the order of integration of the variables reflecting causality. Table 5 presents information on causalities that are integrated of order 1, as both tests confirmed the non-stationarity of levels and the stationarity of first differences. Additionally, it includes causalities that may be integrated of order 1, where one test indicated non-stationarity at the level and stationarity at the first difference, while the other test indicated stationarity at the level.<sup>8</sup>

Causality	Lags	Order of integration
From the U.S. stock market to the DE stock market	2	I(0) or I(1)
From the U.S. bond market to the DE bond market	2	I(0) or I(1)
From DE bond market to the U.S. bond market	1	I(1)
From volatility of the DE bond market to volatility of the U.S. bond market	9	I(0) or I(1)

Table 5. Measures of causality, which turned out to be integrated of order 1 or may be integrated of order 1

In addition to understanding the order of integration of causality measures, it is essential to determine the order of integration for the variables used as determinants of causality. Table 6 presents the results of tests conducted to ascertain the order of integration for these factors.

The results presented in Table 6 indicate that certain time series of exchange rates may be generated by stochastic processes integrated of order 1. The primary balance for the United States is identified as a trend-stationary variable. For variables identified as I(1), we treat them accordingly. In the case of non-stationary causality measures, we seek to identify long-run

<sup>&</sup>lt;sup>8</sup> For causalities not listed in Table 5, the tests indicated stationarity.

relationships between causality and variables reflecting exchange rates. Subsequently, we estimate the parameters of an error correction model to explain changes in causality using stationary variables. When causality measures are stationary, we estimate the parameters of a standard linear regression model with stationary regressors, utilizing the first differences of variables associated with exchange rates.

	Le	evel	First di	fference	— Decision	
	ADF statistic	KPSS statistic	ADF statistic	KPSS statistic	Decision	
ESI	-2.65	0.08			I(0)	
1'51	(n,1)	0.08	-	-	1(0)	
MDCECB	-4.60	0.00			I(0)	
MF S <sub>S</sub>	(d,1)	0.09	-	-	1(0)	
MDCFed	-4.94	0.08			I(0)	
MFSs	(n,1)	0.08	-	-	1(0)	
$CAPB_{s}^{DE}$	-4.20	0.22	-	-	I(0)	
CADDUS	-6.38	0.11			I(0), after	
CAPDS	(t,1)	0.11	-	-	detrending	
DFFDDE	-2.97	0.00	-3.90		I(0)  or  I(1)	
KEEK <sub>S</sub>	(d,1)	0.09	(n,0)	-	1(0) 01 1(1)	
DEEDUS	0.92	0.43	-2.37		I(0)  or  I(1)	
<u><u><u>K</u></u><u>L</u><u>L</u><u>K</u><u>S</u></u>	(n,1)	0.45	(n,0)	-		
EVDATEUSD/EUR	-2.33	0 10	-2.96		I(0)  or  I(1)	
EXKAIE	(d,1)	0.10	(n,0)	-	I(0)  of  I(1)	

Table 6. Testing order of integration of variables considered as determinants of causalities

Notes: t,d,n indicates that a model with trend, with drift and without drift was used in regression of the ADF test.

	Table 7. Econom	cu ic moucis expiai	ning time-var	ying causanty
Model	Markets (direction)	Countries (direction)	Lag	Short-run (S) or long-run (L) equation
1	$S \rightarrow S$	$US \rightarrow DE$	1	S
2	$B \rightarrow B$	$US \rightarrow DE$	1	S
3	$B \rightarrow B$	$US \rightarrow DE$	3	S
4	$B \rightarrow B$	$DE \rightarrow US$	1	S & L
5	$B \rightarrow B$	$DE \rightarrow US$	2	S
6	$B \rightarrow S$	DE	1	S
7	$S \rightarrow B$	DE	1	S
8	$S \rightarrow B$	DE	3	S
9	$B \rightarrow S$	US	1	S
10	$S \rightarrow B$	US	3	S
11	$B \rightarrow S$	US	1	S
12	$B \rightarrow S$	US	3	S
13	$S \rightarrow S$ (Vol.)	$US \rightarrow DE$	9	S
14	$B \rightarrow S$ (Vol.)	DE	9	S

Table 7. Econometric models explaining time-varying causality

Notes: *S*, *B*, and Vol. stand for Stocks, Bonds and volatility, respectively.

Model	M1	M2	M3	M4	M5	M6	M7	M8	M9	M10	M11	M12	M13	M14
Markets	$S \rightarrow S$	$B \rightarrow B$	$B \rightarrow B$	$B \rightarrow B$	$B \rightarrow B$	$B \rightarrow S$	$S \rightarrow B$	$S \rightarrow B$	$B \rightarrow S$	$S \rightarrow B$	$B \rightarrow S$	$B \rightarrow S$	$S \rightarrow S$ (Vol.)	$B \rightarrow S$ (Vol.)
Direction (country)	$US \rightarrow DE$	$US \rightarrow DE$	$US \rightarrow DE$	$DE \rightarrow US$	$DE \rightarrow US$	DE	DE	DE	US	US	US	US	$US \rightarrow DE$	DE
Lag	1	1	3	1	2	1	1	3	1	3	1	3	9	9
const	-	-	-	-0.311 (0.216)	-	-	-	-	-	-	-	-	-	-
$EXRATE_s^{USD/EUR}$	-	-	-	0.457* (0.252)	-	-	-	-	-	-	-	-	-	-
CIDF statistic	-	-	-	-5.350***	-	-	-	-	-	-	-	-	-	-
ECT	-	-	-	-1.263*** (0.213)	-	-	-	-	-	-	-			
const	0.072*** (0.019)	0.056*** (0.011)	0.017 (0.012)	-	-	-0.048*** (0.014)	-0.008 (0.011)	-0.019 (0.013)	0.010 (0.014)	-0.006 (0.011)	0.002 (0.015)	-0.019 (0.011)	0.055* (0.032)	0.072* (0.039)
FSI	-	-	-	-	-	-	-0.019** (0.007)	-0.019** (0.009)	-	-	-0.021** (0.010)	-	0.089*** (0.022)	0.053* (0.027)
MPS <sup>ECB</sup>	0.232* (0.111)	-	-	-	0.139* (0.074)	-	-	-	-	-0.138* (0.067)	-	-	-	-
$MPS_s^{Fed}$	-	-	-	-	-	-0.094** (0.041)	-	-	0.103** (0.040)	-	-	-	-	-
$CAPB_s^{DE}$	-0.032* (0.018)	-	-	-	-	-	-	-	-	0.026** (0.011)	-	-	-	-
$CAPB_{s}^{US}$	-	-0.019** (0.009)	-	-	0.021* (0.011)	-	-	0.019** (0.009)	-	-	-	-	-	-
$\Delta EXRATE_s^{USD/EUR}$	-	-	-0.322* (0.171)	-	-	-	0.284** (0.135)	-	-	-	-	0.343** (0.150)	-	-
R-squared	0.311	0.172	0.151	0.626	0.337	0.201	0.266	0.326	0.238	0.534	0.178	0.206	0.442	0.156

Table 8. Results of the estimation of parameters of regression models explaining causalities

Notes: const are the constant,  $REER_s^{DE}$  and  $REER_s^{US}$  are the Real Effective Exchange rate of Germany and the US, respectively,  $EXRATE_s^{USD/EUR}$  is the bilateral exchange rate between USD and EUR. CIDF statistic is used for testing whether cointegration occurs and, ECT is the error correction term. FSI is the Financial Stress indicator,  $MPS_s^{ECB}$  and  $MPS_s^{Fed}$  are the Monetary policy shocks of Germany and the US, respectively,  $CAPB_s^{DE}$  and  $CAPB_s^{US}$  are the Primary balance of Germany and the US, respectively,  $\Delta EXRATE_s^{USD/EUR}$  is the change in the bilateral exchange rate between USD and EUR. \*,\*\*,\*\*\* denote the level of significance of 10%, 5% and 1% levels, respectively. In parentheses are the standard deviation.

With the purpose of explaining causality, Table 7 summarizes the fourteen models we utilize to analyze the impact of financial stress index, monetary policy shocks, fiscal primary balance and exchange rate on cross-quantilogram predictability, while Table 8 reports the estimation parameters of the regression models.

Examining Table 8, the cointegrating relation results indicate that during periods of a stronger U.S. dollar-euro exchange rate,  $EXRATE_s^{USD/EUR}$ , changes in the 10-yearU.S.sovereign bond yields to increases of the 10-year sovereign bond yields in Germany was more intensive (Model 4 of the Long-run equation). In other words, the reactions of theU.S.10-year bond yields to changes in 10-year German bond yields at the 0.95 quantile of both variables are stronger when the euro is weaker. This finding suggests an inverted uncovered interest rate parity consistent with Czudaj and Prüser (2015) who also identified a bidirectional relationship between long-term interest rates and exchange rate for Germany and the United States. However, prior empirical studies (Czudaj and Prüser, 2015) have shown that U.S. economic variables exert a stronger influence on Germany than vice versa.

In the short term, exchange rate fluctuations also play a moderating role. Models 3 and 7 show that as the euro appreciates against the U.S. dollar, causality from the U.S. bond market to the German bond market increases. Conversely, when the euro weakens, causality from the German stock market to the German bond market intensifies. This suggests that exchange rate dynamics influence the interplay between Germany's stock and bond markets.

Further, the European Central Bank monetary policy shocks ( $MPS_s^{ECB}$ ) emerge as critical drivers of causality between German and U.S. markets (Models 1, 5, and 10). For instance, the negative reaction of Germany rates of return to a significant decline of the S&P 500 is more pronounced during periods of positive monetary policy shocks (Models 1). This result aligns with the results present in Table 2 during the 2006-2008 period and with Kazi et al. (2013), who noted that the impact of monetary policy shocks varies over time and international propagation of shocks is more intensive during turbulent periods. Monetary policy shocks of the European Central Bank also report a significant impact on causality between the German bond market and the U.S. bond market (Model 5). Indicating that raising the reference rate resulted in significant increase of 10-year sovereign bond yields in Germany, which later transmitted to the U.S. bond market. We also observe that the U.S. treasury bond market adjusted to the tightening of monetary policy in the euro zone in Model 10.

In contrast, the U.S. Federal Reserve monetary policy shocks  $(MPS_s^{Fed})$  affect causality within the domestic market. Specifically, Model 9 shows that a lower-than-expected rising Fed reference

rate reduces causality from the U.S. bond market to the U.S. stock market (positive estimate coefficient of 0.103). This finding aligns with Miranda-Agrippino and Ricco (2021), who demonstrated that a monetary tightening in the U.S. is contractionary, with deterioration of domestic demand, labour, prices of assets and agents' expectations.

The fiscal primary balance also significantly impacts cross-market dynamics. In Germany, a higher primary balance strengthens the resilience of the German stock market to negative returns in the U.S. stock market (Model 1). This result, derived from data spanning 1990-2024, contrasts with earlier studies (e.g. Afonso and Sousa, 2011), which found that fiscal shocks had a minimal impact on Germany's asset markets prior to 2008. However, the role of fiscal policy has grown substantially since the Global Financial Crisis and subsequent crises. In the U.S., fiscal policy behaves differently. Model 5 reveals that a higher primary balance does not improve the resilience of the U.S. bond market to negative shocks from Germany, a counterintuitive result.

The financial stress indicator also demonstrates consistent significance across Models 7, 8, 11 and 13-14. During periods of heightened financial stress, causality from the German stock market to the German bond market decreases (Models 7 and 8), suggesting that German sovereign bonds act as a safe haven. This finding aligns with Dajcman (2012), who observed that during periods of market volatility, the correlation between stock market returns and sovereign bond yield dynamics differed between Germany and countries affected by the sovereign debt crisis. Further, the author showed that in Germany this correlation was predominantly positive, whereas in the affected countries, it was largely negative.

Lastly, for the US, financial stress impacts causality between its bond and stock markets in more complex ways. Model 11 reveals that using a one-day lag, impact of financial stress on causality turned out to be negative. Moreover, in periods of higher financial stress, causality in volatility intensifies, as expected (Models 13 and 14).

## 4.3. Robustness Analysis

To have a clear view of the predictability relationship between stocks and bonds in Germany and the United States, in this subsection we explore the causality interdependence between these two markets considering quantile of order 0.95 for stocks and 0.05 for bonds (Tables A1 and A2 in the Appendix). Further, we analyze the change in volatility in stock and bonds for quantile of order 0.05, in Table A3 of the Appendix.

During the 1990s (Table A1), the aftermath of the Gulf War and German reunification led to negative causality from U.S. to German highest quantiles in stock markets. However, Germany's

economic recovery positively influenced the U.S. stocks. Similar trends persisted throughout the 1990s, influenced by events such as the Maastricht Treaty and European integration, which created uncertainties and adjustments in German markets while highlighting the growing influence of U.S. financial dynamics on Germany.

During the 21<sup>st</sup> century, mutual positive causality emerged in the highest quantiles of stocks, particularly after the dot-com bubble and the September 11 attacks, underscoring the high interdependence of the markets in times of crisis. This pattern continued during the 2007–2009 global financial crisis, where shocks originating in the U.S. had profound global effects. The positive causality in both directions highlighted the interconnectedness of financial systems and the role of coordinated monetary and fiscal policy responses. Similar trends were observed during subsequent crises, including the European sovereign debt crisis from 2011 to 2013, the Brexit referendum, and the U.S. presidential election from 2016 to 2018, where global uncertainties heightened market interdependencies. The COVID-19 pandemic further reinforced the bilateral causality between markets, reflecting their global nature and the synchronized policy responses to mitigate volatility and risks.

Parallel analyses of bond yields revealed similar interdependencies. Before the 21st century, the majority of the relationship between the two markets is negative. For instance, changes in the lowest quantiles of bond yields in the U.S. typically affect in a opposite direction changes in German bonds. During crises, such as the 2008 financial crises U.S. monetary policy and macroeconomic stability often influenced positively German bond markets. Conversely, Germany's robust economic performance in certain periods (for example, the Covid 19 period) also impacted the U.S. bond yields, illustrating a two-way connection, which confirm the causal relationship observed in Table 2. Additionally, in Germany and in the U.S. (Table A2 in the Appendix), there is a clear increase in the predictability of cross assets after the 2008 financial crisis.

The volatility analysis (Table A3 in the Appendix) further showed significant mutual influences, especially for the lowest volatilities of stocks. When volatility decreases in one stock market, it signals stability, encouraging similar behavior in connected markets. For the bond market, further corroborates this financial integration, as movements from the U.S. and from Germany display similar signals and behavior throughout the year. Further, in Germany during the global financial crisis and in the COVID-19 pandemic a reciprocal transmission of shocks and the critical role of coordinated policy measures in stabilizing financial systems. This pattern is also clear in the U.S. for the 2000s and the late 2010s.

# 5. Concluding Remarks and Policy Implications

In this study, we address a significant gap in the literature by utilizing a novel methodological approach to examine the stock-sovereign bond relationships in the U.S. and Germany over the period of 1990-2024. Through the application of the cross-quantilogram method, the analysis provides a nuanced understanding of these relationships, particularly during periods of financial market stress. By offering a comprehensive examination of both within country and cross-country dynamics, this research aims to advance our understanding of financial market behavior and provides meaningful insights to both academic research and practical policymaking.

Specifically, we analyze directional predictability across various lags (ranging from one to three periods) and different sub-periods, assessing the predominant direction of causality, i.e., from Germany to the United States or vice versa. When analyzing the same asset (sovereign bonds or stocks) or within countries, we examined the direction of the predictability from stocks to bonds and vice versa. These analyses encompass both rates of return (yield changes) and the volatilities of stock and bond markets. To the best of our knowledge, this is the first study to apply this methodology to the problem of causality in variance. Lastly, we examine the impact of financial stress, monetary policy shocks, fiscal stance and exchange rates on causality in extreme quantiles of distribution of rates of return, changes in sovereign bond yields, stock market volatilities and bond volatilities. This analysis applies the error correction model for non-stationary causalities and standard linear regression model for stationary causalities.

Firstly, our analysis reveals significant patterns in the cross-relationships between stock and bond markets in the U.S. and Germany. In the early 1990s, extreme negative stock returns in the U.S. were followed by similar movements in Germany, while negative returns in Germany led to opposite movements in the U.S. This can be attributed to differing economic conditions and recovery trajectories. The period from 2006 to 2008 showed heightened interconnectedness during the global financial crisis, and from 2018 to 2020, strong bidirectional spillovers were observed, influenced by geopolitical events like the US-China trade war and Brexit.

In the bond markets, the European sovereign debt crisis highlighted significant yield spikes and contagion effects, emphasizing the mutual transmission of financial shocks between the U.S. and Germany. Within Germany, the bidirectional relationship between bonds and stocks was positive and significant until the early 21<sup>st</sup> century, becoming more variable post-2000 with the introduction of the Euro. Particularly, during the sovereign debt crisis, heightened market volatility and risk-averse investor behavior caused stocks and bonds to move in opposite directions as investors

shifted capital to manage risk. Similarly, in the U.S., the early 2000s showed positive bilateral predictability between stock and bond returns, with the 2010s marked by significant cross-asset predictability. In both the U.S. and in Germany markets, we conclude that during financial crises periods, volatility events in one market led to a synchronized movement in the other.

Secondly, we devote our analysis to the determinants of the causal bilateral predictability between the U.S. and Germany. We conclude that in the short term, exchange rate fluctuations play a moderating role. When the euro appreciates against the U.S. dollar, causality from the U.S. bond market to the German bond market increases. Conversely, when the euro weakens, causality from the German stock market to the German bond market intensifies, indicating that exchange rate dynamics influence the interplay between Germany's stock and bond markets.

We observe a negative reaction of Germany returns to a significant decline of the S&P 500, which was more pronounced during periods of positive monetary policy shocks. European Central Bank Monetary policy shocks significantly impact the causality between German bond market and U.S. bond market. Thus, raising the reference rate leads to increased sovereign bond yields in Germany, which then transmitted to the U.S. bond market. The U.S. bond market also adjusts to tightening monetary policy in the euro zone. In addition, a lower-than-expected rising Fed reference rate reduces causality from the U.S. bond market to the U.S- stock market.

Fiscal primary balance significantly impacts cross-market dynamics. In Germany, a higher primary balance strengthens the resilience of the German stock market to negative returns in the U.S. stock market. In the U.S., however, a higher primary balance does not improve the resilience of the U.S. bond market to negative shocks from Germany.

Lastly, we consistently find statistical significance for the financial stress indicator. Hence, during periods of heightened financial stress, causality from the German stock market to the German bond market decreases, suggesting that German sovereign bonds act as a safe haven in a flight-to-safety context. In the U.S., financial stress impacts causality between its bond and stock markets negatively for a one-day lag. Moreover, during periods of higher financial stress, causality in volatility intensifies, as expected.

Our findings highlight the importance of policymakers and investors to understand the dynamic interactions between the U.S, and Germany to anticipate and mitigate the impact of global financial shocks. Therefore, our study reveals a complex network of causality influenced by historical events, macroeconomics variables, and policy coordination, demonstrating the profound interconnection between these two major markets. Future research could extend this analysis to other markets.

# References

1. Abakah, E. J. A., A. K. Tiwari, C.-C. Lee, & Ntow-Gyamfi, M. (2023). Quantile price convergence and spillover effects among Bitcoin, Fintech, and artificial intelligence stocks. *International Review of Finance*, 23, 187-205. <u>https://doi.org/10.1111/irfi.12393</u>

2. Adrian, T., Crump, R. K., Vogt, E. (2019). Nonlinearity and Flight-to-Safety in the Risk-Return Trade-Off for Stocks and Bonds. *The Journal of Finance*, 74(4), 1931-1973. <u>https://doi.org/10.1111/jofi.12776</u>

3. Afonso, A., Alves, J., & Monteiro, S. (2024a). Sovereign risk dynamics in the EU: the time varying relevance of fiscal and external (im) balances. *Journal of International Financial Markets, Institutions & Money, 94.* <u>https://doi.org/10.1016/j.intfin.2024.102026</u>

4. Afonso, A., Mignon, V., Saadaoui, J. (2024b), On the time-varying impact of China's bilateral political relations on its trading partners: doux commerce or trade follows the flag?, *China Economic Review*, 85, 102184. https://doi.org/10.1016/j.chieco.2024.102184

5. Afonso, A., & Sousa, R. M. (2011). What are the effects of fiscal policy on asset markets? *Economic Modelling*, *28*(4), 1871-1890. <u>https://doi.org/10.1016/j.econmod.2011.03.018</u>

6. Amemiya, T. (1985). Advanced Econometrics. Harvard University Press.

7. Andersson, M., Krylova, E., & Vähämaa, S. (2007). Why does the correlation between stock and bond returns vary over time? *Applied Financial Economics*, *18*(2), 139-151. https://doi.org/10.1080/09603100601057854

8. Andersen, T. G., Bollerslev, T., Diebold, F. X., & Vega, C. (2007). Real-time price discovery in global stock, bond and foreign exchange markets. *Journal of International Economics*, 73(2), 251-277. <u>https://doi.org/10.1016/j.jinteco.2007.02.004</u>

9. Ardagna, S. (2009). Financial markets' behavior around episodes of large changes in the fiscal stance. *European Economic Review*, 53(1), 37-55. https://doi.org/10.1016/j.euroecorev.2008.07.003

10. Baker, M., & Wurgler, J. (2007). Investor sentiment in the stock market. Journal of *Economic Perspectives*, 21(2), 129-151. <u>https://doi.org/10.1257/jep.21.2.129</u>

11. Barsky, R. B. (1989). Why Don't the Prices of Stocks and Bonds Move Together? *The American Economic Review*, 79(5), 1132-1145.

12. Baruník, J., & Křehlík, T. (2018). Measuring the frequency dynamics of financial connectedness and systemic risk. *Journal of Financial Econometrics*, *16*(2), 271-296. https://doi.org/10.1093/jjfinec/nby001 13. Baur, D. G., & Lucey, B. M. (2009). Flights and contagion—An empirical analysis of stock–bond correlations. *Journal of Financial Stability*, 5(4), 339-352. https://doi.org/10.1016/j.jfs.2008.08.001

14. Beber, A., Brandt, M., & Kavajecz, K. (2009). Flight-to-Quality or Flight-to-Liquidity? Evidence from the Euro-Area Bond Market. *The Review of Financial Studies*, *22*(3), 925-957. https://doi.org/10.1093/rfs/hhm088

15. Bekaert, G., Ehrmann, M., Fratzscher, M., & Mehl, A. (2014). The Global Crisis and Equity Market Contagion. *The Journal of Finance*, 69(6), 2597-2649. https://doi.org/10.1111/jofi.12203

16. Bernanke, B. S., & Kuttner, K. N. (2005). What Explains the Stock Market's Reaction to Federal Reserve Policy? *The Journal of Finance*, 60(3), 1221-1257. https://doi.org/10.1111/j.1540-6261.2005.00760.x

17. Bodart, V., & Reding, P. (1999). Exchange rate regime, volatility and international correlations on bond and stock markets. *Journal of International Money and Finance*, *18*(1), 133-151. <u>https://doi.org/10.1016/S0261-5606(98)00042-4</u>

 Bonfiglioli, A., & Favero, C. A. (2005). Explaining co-movements between stock markets: The case of US and Germany. *Journal of International Money and Finance*, *24*(8), 1299-1316. <u>https://doi.org/10.1016/j.jimonfin.2005.08.016</u>

19. Brière, M., Chapelle, A., & Szafarz, A. (2012). No contagion, only globalization and flight to quality. *Journal of International Money and Finance*, *31*(6), 1729-1744. https://doi.org/10.1016/j.jimonfin.2012.03.010

20. Campbell, J. Y., Pflueger, C., & Viceira, L. M. (2020). Macroeconomic Drivers of Bond and Equity Risks. *Journal of Political Economy*, *128*(8), 3148-3185. https://doi.org/10.1086/707766

21. Centinaio, A., Pacicco, F., Serati, M., & Venegoni, A. (2024). Government decisions and macroeconomic stability: Fiscal policies and financial market fluctuations. *International Review of Economics & Finance*, *96*(A), 103599. <u>https://doi.org/10.1016/j.iref.2024.103599</u>

22. Chen, S. (2007). Does Monetary Policy Have Asymmetric Effects on Stock Returns? *Journal of Money, Credit and Banking*, *39*(2-3), 667-688. <u>https://doi.org/10.1111/j.0022-</u>2879.2007.00040.x

23. Chiang, T. C., Li, J., & Yang, S. (2015). Dynamic stock-bond return correlations and financial market uncertainty. *Review of Quantitative Finance and Accounting*, 45(1), 59-88. <u>https://doi.org/10.1007/s11156-013-0430-4</u> 24. Chang H.-W., Change, T., & Wang, M.-C. (2024). Revisit the impact of exchange rate on stock market returns during the pandemic period. *North American Journal of Economics and Finance*, 70, 102068. <u>https://doi.org/10.1016/j.najef.2023.102068</u>

25. Chavleishvili, S., & Kremer, M. (2023). *Measuring Systemic Financial Stress and its Risks for Growth*. ECB Working Paper No. 2023/2842.

26. Cho, Y., Linton, O., & Whang, Y. (2007). Are there Monday effects in stock returns: A stochastic dominance approach. *Journal of Empirical Finance*, *14*(5), 736-755. https://doi.org/10.1016/j.jempfin.2007.02.001

27. Christoffersen, P., & Diebold, F. X. (2002). *Financial Asset Returns, Market Timing, and Volatility Dynamics*. Available at: <u>http://dx.doi.org/10.2139/ssrn.306127</u>

28. Chuliá, H., & Torro, H. (2008). The economic value of volatility transmission between the stock and bond markets. *Journal of Futures Markets: Futures, Options, and Other Derivative Products*, 28(11), 1066-1094. <u>https://doi.org/10.1002/fut.20342</u>

29. Connolly, R., Stivers, C., & Sun, L. (2005). Stock Market Uncertainty and the Stock-Bond Return Relation *Journal of Financial and Quantitative Analysis*, 40(1), 161-194. https://doi.org/10.1017/S0022109000001782

30. Cowles, A., & Jones, H. (1937). Some A Posteriori Probabilities in Stock Market Action, *Econometrica*, 5, 280-294. <u>https://doi.org/10.2307/1905515</u>

31. Czudaj, R., & Prüser, J. (2015). International parity relationships between Germany and the USA revisited: evidence from the post-DM period. *Applied Economics*, 47(26), 2745-2767. https://doi.org/10.1080/00036846.2015.1008776

32. Dai, Z., Zhu, J., & Zhang, X. (2022). Time-frequency connectedness and cross-quantile dependence between crude oil, Chinese commodity market, stock market and investor sentiment. *Energy Economics*, *114*, 106226. <u>https://doi.org/10.1016/j.eneco.2022.106226</u>

 Davis, R. A., & Mikosch, T. (2009). The extremogram: A correlogram for extreme events. Bernoulli, 15(4), 977-1009. <u>https://doi.org/10.3150/09-BEJ213</u>

34. Davis, R. A., Mikosch, T., & Cribben, I. (2012). Towards estimating extremal serial dependence via the bootstrapped extremogram. *Journal of Econometrics*, *170*(1), 142-152. https://doi.org/10.1016/j.jeconom.2012.04.003

35. Davis, R. A., Mikosch, T., & Zhao, Y. (2013). Measures of serial extremal dependence and their estimation. *Stochastic Processes and their Applications*, *123*(7), 2575-2602. https://doi.org/10.1016/j.spa.2013.03.014 36. Dean, W. G., Faff, R. W., & Loudon, G. F. (2010). Asymmetry in return and volatility spillover between equity and bond markets in Australia. *Pacific-Basin Finance Journal*, *18*(3), 272-289. https://doi.org/10.1016/j.pacfin.2009.09.003

37. Dickey, D. A., & Fuller, W. A. (1979). Distribution of the Estimators for Autoregressive Time Series With a Unit Root. *Journal of the American Statistical Association*, 74(366), 427-431. https://doi.org/10.2307/2286348

38. Dickey, D. A., & Fuller, W. A. (1981). Likelihood Ratio Statistics for Autoregressive Time Series with a Unit Root. *Econometrica*, 49(4), 1057-1072. <u>https://doi.org/10.2307/1912517</u>

39. Diebold, F. X., & Yilmaz, K. (2012). Better to give than to receive: Predictive directional measurement of volatility spillovers. *International Journal of forecasting*, 28(1), 57-66. https://doi.org/10.1016/j.ijforecast.2011.02.006

40. Dajcman, S. (2012). Comovement between stock and bond markets and the 'flight-toquality'during financial market turmoil–a case of the Eurozone countries most affected by the sovereign debt crisis of 2010–2011. *Applied Economics Letters, 19*(17), 1655-1662. <u>https://doi.org/10.1080/13504851.2011.652771</u>

41. Döpke, J., & Pierdzioch, C. (2006). Politics and the stock market: Evidence from Germany. *European Journal of Political Economy*, 22(4), 925-943. <u>https://doi.org/10.1016/j.ejpoleco.2005.11.004</u>

42. Duffee, G. (2023). Macroeconomic News in Asset Pricing and Reality. *The Journal of Finance*, 78(3), 1499-1543. <u>https://doi.org/10.1111/jofi.13218</u>

43. Dufour, A., Stancu, A., & Varotto, S. (2017). The equity-like behaviour of sovereign bonds. *Journal of International Financial Markets, Institutions and Money*, 48, 25-46. <u>https://doi.org/10.1016/j.intfin.2016.11.014</u>

44. Dufour, J. M., Hallin, M., & Mizera, I. (1998). Generalized Runs tests for heteroscedastic time series. *Nonparametric Statistics*, *9*, 39-86. <u>https://doi.org/10.1080/10485259808832735</u>

45. Ferrer, R., Bolós, V. J., & Benítez, R. (2016). Interest rate changes and stock returns: A European multi-country study with wavelets. *International Review of Economics & Finance*, 44, 1-12. <u>https://doi.org/10.1016/j.iref.2016.03.001</u>

46. Ferrer, R., Shahzad, S. J. H., & Maizonada, A. (2019). Nonlinear and extreme dependence between long-term sovereign bond yields and the stock market: A quantile-on-quantile analysis. *Economics Bulletin*, *39*(2), 969-981.

47. Fidora, M., Fratzscher, M., Thimann, C. (2007). Home bias in global bond and equity markets: The role of real exchange rate volatility. *Journal of International Money and Finance*, 26(4), 631-655. <u>https://doi.org/10.1016/j.jimonfin.2007.03.002</u>

48. Flannery, M. J., & Protopapadakis, A. A. (1988). From T-Bills to Common Stocks: Investigating the Generality of Intra-Week Return Seasonality. *The Journal of Finance*, 43(2), 431-450. <u>https://doi.org/10.1111/j.1540-6261.1988.tb03948.x</u>

49. Forbes, K. J., & Rigobon, R. (2002). No Contagion, Only Interdependence: Measuring Stock Market Comovements. *The Journal of Finance*, 57(5), 2223-2261. https://doi.org/10.1111/0022-1082.00494

50. Grabowski, W., & Welfe, A. (2016). An Exchange Rate Model with Market Pressures and a Contagion Effect. *Emerging Markets Finance and Trade*, 52(12), 2706–2720. https://doi.org/10.1080/1540496X.2016.1216931

51. Grabowski, W., & Welfe, A. (2020). The Tobit cointegrated vector autoregressive model: An application to the currency market. *Economic Modelling*, *89*, 88-100. <u>https://doi.org/10.1016/j.econmod.2019.10.008</u>

52. Gerrits, R., & Yuce, A. (1999). Short- and long-term links among European and US stock markets. *Applied Financial Economics*, *9*(1), 1-9. <u>https://doi.org/10.1080/096031099332483</u>

53. Grammatikos, T., & Vermeulen, R. (2012). Transmission of the financial and sovereign debt crises to the EMU: Stock prices, CDS spreads and exchange rates. *Journal of International Money and Finance*, *31*(3), 517-533. <u>https://doi.org/10.1016/j.jimonfin.2011.10.004</u>

54. Han, H., Linton, O., Oka, T., & Whang, Y.-J. (2016). The cross-quantilogram: Measuring quantile dependence and testing directional predictability between time series. *Journal of Econometrics*, *193*, 251-270. https://doi.org/10.1016/j.jeconom.2016.03.001

55. Ilmanen, A. (2003). Stock-Bond Correlations. *The Journal of Fixed Income*, *13*(2), 55-66. https://doi.org/10.3905/jfi.2003.319353

56. Jammazi, R., Ferrer, R., Jareño, F., & Hammoudeh, S. M. (2017). Main driving factors of the interest rate-stock market Granger causality. *International Review of Financial Analysis*, *52*, 260-280. <u>https://doi.org/10.1016/j.irfa.2017.07.008</u>

57. Jammazi, R., Tiwari, A., Ferrer, R., & Moya, P. (2015). Time-varying dependence between stock and government bond returns: International evidence with dynamic copulas. *North American Journal of Economics and Finance*, 33, 74-93. <u>https://doi.org/10.1016/j.najef.2015.03.005</u>

58. Jarociński, M., & Karadi, P. (2020). Deconstructing Monetary Policy Surprises—The Role of Information Shocks. *American Economic Journal: Macroeconomics*, *12*(2), 1-43. https://doi.org/10.1257/mac.20180090

59. Johnson, R. and Soenen, L. (2009). European Economic Integration and Stock Market Co-Movement with Germany. *Multinational Business Review*, 17, 3, 205-228. <u>https://doi.org/10.1108/1525383X200900024</u>

60. Karim, S., Naeem, M. A., Shafiullah, M., Lucey, B. M., & Ashraf, S. (2023). Asymmetric relationship between climate policy uncertainty and energy metals: Evidence from cross-quantilogram. *Finance Research Letters*, *54*. 103728. <u>https://doi.org/10.1016/j.frl.2023.103728</u>

61. Kazi, I. A., Wagan, H., & Akbar, F. (2013). The changing international transmission of US monetary policy shocks: Is there evidence of contagion effect on OECD countries. *Economic Modelling*, 30, 90-116. https://doi.org/10.1016/j.econmod.2012.07.020

62. Kim, S., & In, F. (2007). On the relationship between changes in stock prices and bond yields in the G7 countries: Wavelet analysis. *Journal of International Financial Markets, Institutions and Money*, *17*(2), 167-179. <u>https://doi.org/10.1016/j.intfin.2005.10.004</u>

63. Kumar, S., Khalfaoui, R., & Tiwari, A. K. (2021). Does geopolitical risk improve the directional predictability from oil to stock returns? Evidence from oil-exporting and oil-importing countries. *Resources Policy*, *74*, 102253. <u>https://doi.org/10.1016/j.resourpol.2021.102253</u>

64. Kwiatkowski, D., Phillips, P. C. B., Schmidt, P., & Shin, Y. (1992). Testing the null hypothesis of stationarity against the alternative of a unit root: How sure are we that economic time series have a unit root? *Journal of Econometrics*, 54(1-3), 159-178. https://doi.org/10.1016/0304-4076(92)90104-Y

65. Lakdawala, A., Moreland, T., & Schaffer, M. (2021). The international spillover effects of US monetary policy uncertainty. *Journal of International Economics*, 133, 103525. <u>https://doi.org/10.1016/j.jinteco.2021.103525</u>

66. Lee, B. (1992). Causal Relations Among Stock Returns, Interest Rates, Real Activity, and Inflation. *The Journal of Finance*, 47(4), 1591-1603. <u>https://doi.org/10.1111/j.1540-6261.1992.tb04673.x</u>

67. Lee, H. (2021). Time-varying comovement of stock and treasury bond markets in Europe: A quantile regression approach. *International Review of Economics & Finance*, 75, 1-20. <u>https://doi.org/10.1016/j.iref.2021.03.020</u> 68. Linton, O., & Whang, Y. J. (2007). The quantilogram: With an application to evaluating directional predictability. *Journal of Econometrics*, *141*(1), 250-282. https://doi.org/10.1016/j.jeconom.2007.01.004

69. Longin, F., & Solnik, B. (2002). Extreme Correlation of International Equity Markets. *The Journal of Finance*, *56*(2), 649-676. <u>https://doi.org/10.1111/0022-1082.00340</u>

70. Maddala, G. S. (1983). *Limited-Dependent and Qualitative Variables in Econometrics*. Cambridge University Press. <u>https://doi.org/10.1017/CBO9780511810176</u>

71. Miranda-Agrippino, S., & Ricco, G. (2021). The Transmission of Monetary Policy Shocks.AmericanEconomicJournal:Macroeconomics,13(3),74-107.https://doi.org/10.1257/mac.20180124

72. Nguyen, H., & Javed, F. (2023). Dynamic relationship between Stock and Bond returns: A GAS MIDAS copula approach. *Journal of Empirical Finance*, 73, 272-292. https://doi.org/10.1016/j.jempfin.2023.07.004

73. Pham, L. (2021). Frequency connectedness and cross-quantile dependence between green bond and green equity markets. *Energy Economics*, 98, 105257. <u>https://doi.org/10.1016/j.eneco.2021.105257</u>

74. Politis, D. N., & Romano, J. P. (1994). The Stationary Bootstrap. *Journal of the American Statistical Association*, 89, 1303-1313. <u>https://doi.org/10.1080/01621459.1994.10476870</u>

75. Razzaq, A., Sharif, A., An, H., & Aloui, Ch. (2022). Testing the directional predictability between carbon trading and sectoral stocks in China: New insights using cross-quantilogram and rolling window causality approaches. *Technological Forecasting and Social Change*, *182*, 121846. <u>https://doi.org/10.1016/j.techfore.2022.121846</u>

76. Reinhart, C. M., & Rogoff, K. S. (2008). Is the 2007 US sub-prime financial crisis so different? An international historical comparison. *American Economic Review*, *98*(2), 339-344.

77. Shiller, R. J., & Beltratti, A. (1992). Stock prices and bond yields: Can their comovements be explained in terms of present value models? *Journal of Monetary Economics*, *30*(1), 25-46. https://doi.org/10.1016/0304-3932(92)90042-Z

78. Tiwari, A. K., Cunado, J., Gupta, R., & Wohar, M. E. (2018). Volatility spillovers across global asset classes: Evidence from time and frequency domains. *The Quarterly Review of Economics and Finance*, 70, 194-202. <u>https://doi.org/10.1016/j.qref.2018.05.001</u>

79. Tobin, J. (1958). Liquidity Preference as Behavior Towards Risk. *The Review of Economic Studies*, 25(2), 65-86. <u>https://doi.org/10.2307/2296205</u>

80. Uddin, G. S., Rahman, M. L., Hedstrom, A. & Ahmed, A. (2019). Cross-quantilogrambased correlation and dependence between renewable energy stock and other asset classes. *Energy Economics*, 80, 743-759. <u>https://doi.org/10.1016/j.eneco.2019.02.014</u>

81. Valchev, R. (2020). Bond Convenience Yields and Exchange Rate Dynamics. *American Economic Journal: Macroeconomics*, *12*(2), 124-166. <u>https://doi.org/10.1257/mac.20170391</u>

82. Viceira, L. M. (2012). Bond risk, bond return volatility, and the term structure of interest rates. *International Journal of Forecasting*, 28(1), 97-117. https://doi.org/10.1016/j.ijforecast.2011.02.018

83. Yang, J., Zhou, Y., & Wang, Z. (2009). The stock–bond correlation and macroeconomic conditions: One and a half centuries of evidence. *Journal of Banking & Finance*, *33*(4), 670-680. <u>https://doi.org/10.1016/j.jbankfin.2008.11.010</u>

84. Zhang, J., Chen, X., Wei, Y., & Bai, L. (2023). Does the connectedness among fossil energy returns matter for renewable energy stock returns? Fresh insights from the Cross-Quantilogram analysis. *International Review of Financial Analysis*, 88, 102659. https://doi.org/10.1016/j.irfa.2023.102659

# Appendix

	1	Relationshin	hetween stock	market returns	s of order 0.95	g ~ ,	Relationship between changes in sovereign bond vields of order 0.05					
Window	Lag	<u>-1</u>	La	<del>n-2</del>	I a	7-3	La	<u>a-1</u>	In	<u>7-7</u>	La	<del>7-3</del>
,, <b>, , , , , ,</b> , , , , , , , , , , , ,	$\frac{DUS}{US \to DE}$	$DE \rightarrow US$	$US \rightarrow DE$	$DE \rightarrow US$	$US \rightarrow DE$	$DE \rightarrow US$	$US \rightarrow DE$	$DE \rightarrow US$	$US \rightarrow DE$	$DE \rightarrow US$	$US \rightarrow DE$	$DE \rightarrow US$
1991-1993	-0.054***	0.183***	0.025	0.025	-0.001	0.051	-0.007	0.018	-0.032**	-0.007	-0.032**	-0.007
1992-1994	-0.028***	0.053	-0.028***	0.027	-0.028***	-0.027***	0.052	0.104*	0.051	-0.001	0.025	-0.001
1993-1995	0.001	0.028	0.028	0.028	0.001	0.001	-0.001	0.157***	0.025	0.000	-0.028*	0.027
1994-1996	-0.028***	0.025	-0.028***	-0.001	-0.001	-0.028***	-0.003	0.076	0.023	-0.029***	-0.055***	-0.003
1995-1997	0.000	0.160**	-0.026	-0.026	0.055	-0.026	-0.001	-0.001	0.025	-0.001	0.027	-0.028
1996-1998	0.052	0.078	-0.001	0.052	0.078	0.051	-0.027	-0.001	-0.001	0.052	0.051	-0.028
1997-1999	-0.001	0.052	-0.054***	-0.001	-0.001	0.051	-0.029	0.050	-0.003	0.076	0.049	-0.029**
1998-2000	0.000	0.025	-0.053***	-0.001	0.080	0.025	-0.054***	0.078	0.025	0.051	-0.027	0.000
1999-2001	-0.001	0.025	-0.028***	-0.001	0.000	-0.027***	-0.028	0.078	-0.053***	-0.001	-0.027	-0.001
2000-2002	0.025	0.025	0.025	-0.001	0.025	0.025	0.016	0.090	-0.033***	-0.009	-0.058***	-0.008
2001-2003	0.052	0.052	0.104	0.051	0.078	0.025	0.025	-0.001	-0.027	-0.054***	-0.053***	-0.027
2002-2004	0.078	0.131**	0.157**	0.104*	0.078	0.051	0.022	0.048	-0.029**	-0.004	-0.055***	-0.055***
2003-2005	0.025	0.131***	0.107	0.051	0.082	0.053	-0.008	0.041	-0.009	-0.033***	-0.033***	-0.058***
2004-2006	-0.001	-0.001	-0.054***	0.000	0.104*	0.000	-0.001	-0.001	-0.028	-0.054***	-0.028	-0.001
2005-2007	0.055	0.001	-0.026	-0.053***	0.082	-0.026	0.022	-0.031***	0.023	-0.005	0.049	-0.005
2006-2008	0.160**	0.107***	0.027	0.001	0.053	0.112*	0.102*	0.050	0.154***	-0.003	0.076	0.076*
2007-2009	0.157***	0.052	0.025	0.025	0.104*	0.130**	0.128*	0.024	0.102*	-0.055***	0.050	-0.001
2008-2010	0.157***	0.052	0.025	0.025	0.104*	0.078	0.128**	-0.002	0.102*	-0.029*	0.050	-0.001
2009-2011	0.051	0.080	0.130**	0.053	0.053	0.082*	0.167*	0.017	0.067	-0.007	-0.032***	-0.008
2010-2012	0.051	0.130**	0.078	0.025	0.051	0.051	0.128*	0.024	0.102**	0.049	-0.003	-0.003
2011-2013	0.051	0.157***	0.051	0.051	0.051	0.051	0.128*	0.050	0.049	0.049	-0.003	0.049
2012-2014	0.025	0.052	0.000	0.025	0.080	0.000	-0.029	-0.002	0.024	0.024	-0.003	-0.003
2013-2015	0.025	0.131**	-0.028***	0.000	-0.001	0.001	0.022	-0.055***	0.022	-0.004	0.022	-0.030***
2014-2016	-0.028***	0.157**	-0.001	0.051	-0.054***	0.051	-0.005	-0.030***	0.022	-0.005	-0.004	-0.031***
2015-2017	-0.028***	0.130*	-0.001	0.051	-0.028***	0.104*	-0.005	-0.030***	-0.005	0.020	-0.031***	0.022
2016-2018	0.025	0.078*	0.160**	0.078*	-0.026	0.187***	-0.007	-0.007	-0.006	0.044	0.019	0.071
2017-2019	0.078	0.025	0.107**	-0.028***	0.000	0.078	-0.003	0.076	0.023	0.049	0.023	0.023
2018-2020	0.104	0.104**	0.052	0.183***	0.078	0.157*	0.025	0.104	0.025	0.078*	-0.028	0.025
2019-2021	0.104	0.104	0.080*	0.183***	0.080	0.183**	0.102*	0.154***	0.024	0.050	-0.001	0.050
2020-2022	0.104	0.078**	0.051	0.236***	0.025	0.104	0.102*	0.154**	0.154***	0.076	0.130***	0.078*
2021-2023	0.082	0.001	0.057	0.055	-0.052***	-0.053***	0.073	0.125*	0.128**	0.125**	0.075	-0.004
2022-202402	0.042	-0.022	0.042	0.074	-0.054***	-0.054***	0.042	0.138**	0.106	0.106	0.074	0.010

Table A1. Values of Cross-quantilogram values for 3 years-rolling-windows reflecting relationships between stock market returns in Germany and the United States in quantiles of order 0.95, and relationships between changes in sovereign bond yields in Germany and the United States in quantiles of order 0.05.

			Germ	any		•	United States						
Window	La	g=1	Lag	<i>y=2</i>	La	g=3	Lag	r=1	Lag	y=2	Lag	y=3	
	$B \rightarrow S$	$S \rightarrow B$											
1991-1993	-0.078*	-0.025	-0.052	0.028	-0.025	-0.025	-0.043	0.007	-0.142***	0.007	-0.042	0.007	
1992-1994	-0.078	0.001	-0.025	-0.025	0.001	0.028*	-0.107*	0.001	-0.053	0.001	-0.053	0.001	
1993-1995	-0.133**	0.053***	0.000	-0.027	-0.027	0.000	-0.027	0.027	-0.027	-0.028	0.000	0.026	
1994-1996	-0.128*	0.029***	0.003	-0.049	-0.049	-0.023	-0.104	0.001	0.028**	-0.078	0.028***	0.054***	
1995-1997	0.001	0.000	0.054***	0.053***	0.028**	0.052***	-0.025	-0.053	-0.025	-0.001	0.001	-0.028	
1996-1998	0.027	-0.104*	-0.052	0.001	-0.025	0.001	-0.052	-0.025	-0.052	-0.052	-0.025	-0.130**	
1997-1999	0.001	-0.052	-0.051	-0.025	0.001	0.001	-0.024	-0.050	-0.102*	-0.024	-0.023	-0.076	
1998-2000	-0.027	-0.025	-0.053	-0.025	-0.027	0.000	0.001	-0.051	-0.025	-0.025	0.001	-0.027	
1999-2001	0.001	0.054***	0.054***	0.028**	-0.027	-0.053	-0.025	-0.025	-0.027	0.028	0.027	0.028	
2000-2002	0.003	0.055***	-0.023	-0.023	-0.025	-0.075**	-0.042	-0.018	0.007	0.007	0.008	-0.042	
2001-2003	0.028	0.001	-0.051	0.027	-0.025	-0.080**	-0.052	0.001	0.028	0.001	0.028*	-0.053	
2002-2004	0.001	0.001	-0.052	-0.052	0.001	-0.078	-0.022	0.029***	0.029***	-0.022	0.030***	-0.022	
2003-2005	0.029**	-0.050	-0.078	-0.050	-0.080**	-0.049	0.007	0.032***	-0.018	-0.067*	0.006	-0.017	
2004-2006	0.001	0.001	0.028	0.000	-0.025	-0.027	0.001	-0.025	-0.025	-0.025	0.028*	0.028*	
2005-2007	-0.027	0.027	-0.027	0.000	-0.027	-0.027	0.003	-0.073	-0.104*	-0.022	-0.025	-0.022	
2006-2008	-0.102**	-0.052	0.029***	0.027	-0.049	-0.055	-0.104	-0.053	-0.210***	-0.160**	-0.104	-0.107**	
2007-2009	-0.024	0.002	0.029**	0.029***	-0.050	-0.024	-0.052	-0.052	-0.131*	-0.104	-0.130	-0.080*	
2008-2010	0.002	0.028***	0.029***	0.002	-0.050	0.003	-0.052	-0.052	-0.131*	-0.131**	-0.130	-0.053	
2009-2011	0.006	-0.044	-0.069	0.006	-0.071	0.006	0.028**	-0.024	-0.051	-0.102**	-0.053	0.029***	
2010-2012	-0.102**	-0.128**	-0.076	-0.049	-0.075	-0.049	-0.025	-0.025	-0.104**	-0.130*	-0.078	0.054***	
2011-2013	-0.104*	-0.130*	-0.183***	-0.025	-0.130**	-0.078	-0.050	-0.076**	-0.076	-0.076	-0.075	0.055***	
2012-2014	-0.078	-0.052	-0.027	-0.051	0.027	-0.078*	0.055***	0.029**	-0.024	0.029***	-0.104**	0.003	
2013-2015	0.004	-0.022	-0.099**	0.004	0.004	0.030***	0.028*	0.054***	-0.107**	0.001	-0.136**	-0.025	
2014-2016	-0.020	0.030***	-0.071*	0.03***	-0.020	0.030***	0.028*	0.054***	-0.025	-0.025	-0.130**	-0.025	
2015-2017	-0.048	-0.022	-0.022	0.03***	-0.022	0.030***	0.029***	0.003	-0.023	-0.049	-0.128**	-0.078**	
2016-2018	-0.069*	-0.019	0.031***	-0.020	0.005	-0.022	-0.024	-0.102**	0.055***	-0.107*	-0.023	-0.082*	
2017-2019	-0.024	-0.05	0.001	-0.023	0.028*	0.055***	0.001	0.001	0.001	-0.025	-0.051	-0.025	
2018-2020	-0.052	-0.025	-0.025	0.001	0.028	0.001	-0.104	-0.078	-0.104*	-0.025	-0.130**	-0.078	
2019-2021	-0.025	-0.025	-0.027	0.028	0.027	0.000	-0.154***	-0.076	-0.076	0.002	-0.128**	-0.050	
2020-2022	-0.104*	0.028	-0.051	0.001	0.001	-0.027	-0.050	-0.076	-0.050	-0.024	-0.154***	-0.104*	
2021-2023	-0.080	0.053***	-0.001	0.000	0.026	-0.053	-0.023	-0.023	0.001	0.003	-0.104*	-0.023	
2022-2024Q2	-0.106*	0.054***	0.022	-0.010	0.022	-0.074	-0.010	-0.042	0.022	-0.010	-0.074	-0.010	

Table A2. Values of cross-quantilogram for 3 years-rolling-windows reflecting relationship between stock (quantile of order 0.95) and bond market (quantile of order 0.05) returns in Germany and in the United States.

	stock markets for quanties of order 0.05. (Eag 7)												
Win down	Stock v	s. Stock	Bond v	s. Bond	DE (betwee	en markets)	US (betwee	en markets)					
winaow	$US \rightarrow DE$	$DE \rightarrow US$	$US \rightarrow DE$	$DE \rightarrow US$	$B \rightarrow S$	$S \rightarrow B$	$B \rightarrow S$	$S \rightarrow B$					
1991-1993	-0.054***	-0.028***	-0.054***	-0.028*	0.130	-0.054***	-0.028	-0.054***					
1992-1994	-0.052***	-0.002	-0.026	0.077	0.130	-0.054***	0.001	-0.052***					
1993-1995	-0.028**	-0.002	-0.052***	-0.028**	-0.055***	-0.053***	-0.027	-0.002					
1994-1996	-0.055***	-0.025	-0.024	0.156*	-0.028***	-0.021	-0.028***	0.051					
1995-1997	-0.054***	-0.022	-0.050***	0.066	-0.025	-0.028***	-0.045***	0.059					
1996-1998	-0.028***	0.194*	0.112	0.051	-0.054***	-0.025	-0.052***	-0.028***					
1997-1999	-0.053***	-0.054***	0.139	0.051	0.000	-0.054***	-0.025	0.026					
1998-2000	-0.055***	0.051	0.104	0.025	0.104	-0.055***	-0.028***	0.025					
1999-2001	-0.051***	0.156	-0.028***	0.025	0.104	-0.055***	-0.002	0.033					
2000-2002	-0.055***	0.209**	-0.002	0.053	0.106	-0.055***	-0.002	0.025					
2001-2003	0.010	0.149	0.025	-0.053***	-0.053***	0.064	-0.054***	-0.019					
2002-2004	0.005	0.101	0.032	-0.054***	0.104	0.220*	0.059	-0.051***					
2003-2005	0.087	-0.050***	0.007	-0.054***	-0.028**	0.124	0.035	0.059					
2004-2006	0.035	-0.025	-0.026	-0.002	-0.028***	0.057	0.001	0.263**					
2005-2007	0.025	0.051	0.059	0.130*	-0.028***	-0.024	0.025	0.156					
2006-2008	0.000	0.077	0.084	0.213**	0.026	0.030	0.051	0.080					
2007-2009	0.145	0.232**	0.082	0.186	0.194*	0.061	0.249*	0.057					
2008-2010	0.273*	0.220*	0.130*	0.026	-0.054***	0.130*	-0.002	0.010					
2009-2011	0.209*	0.209**	0.130*	0.025	-0.028***	0.130**	-0.002	0.077					
2010-2012	0.221**	0.130	0.100	-0.026	0.061*	-0.009	-0.053***	-0.028***					
2011-2013	0.149	0.025	0.053	0.004	0.092	-0.055***	-0.054***	0.031					
2012-2014	0.000	-0.028**	0.025	0.000	-0.028**	-0.028**	0.051	0.001					
2013-2015	0.030	-0.002	0.077	0.023	-0.026	-0.029***	0.053	-0.002					
2014-2016	0.051	-0.021	0.055	-0.025	-0.054***	-0.023	0.059	0.001					
2015-2017	0.030	0.026	0.053	-0.054***	-0.027	0.000	-0.027	0.084					
2016-2018	0.025	0.025	-0.055***	-0.055***	-0.055***	-0.002	-0.028**	0.025					
2017-2019	0.030	-0.05***	-0.053***	-0.055***	0.002	-0.002	-0.019	-0.002					
2018-2020	0.262*	0.077	-0.024	-0.054***	0.051	-0.054***	-0.024	0.183**					
2019-2021	0.104	0.025	-0.028**	-0.028**	0.077	-0.054***	-0.054***	0.051					
2020-2022	0.077	0.077	-0.028**	-0.002	0.051	-0.054***	-0.054***	-0.054***					
2021-2023	-0.002	0.057	0.051	0.133	-0.002	-0.053***	0.183*	0.106					
2022-2024O2	-0.054***	0.073*	0.037	0.177**	0.076	-0.047***	0.017	-0.053***					

 Table A3. Values of cross-quantilogram for 3 years-rolling-windows reflecting relationship between volatilities in stock markets for quantiles of order 0.05. (Lag=9)